

LONDON:472977.4148

TABLE OF CONTENTS

1.	DEFINITIONS AND INTERPRETATION	2
2.	EFFECTIVENESS OF THIS DEED AND LONG-STOP TIME	7
3.	CONFIRMATIONS	7
4.	REPRESENTATIONS	13
5.	RELATIONSHIP WITH OTHER DOCUMENTS	16
6.	AGREEMENT TO SUBSCRIBE	16
7.	MUTUAL RELEASES	19
8.	EFFECTIVENESS OF MUTUAL RELEASES	22
9.	TRANSFERS	22
10.	ACCESSION	22
11.	FURTHER ASSURANCES	23
12.	THIRD PARTY RIGHTS	23
13.	WAIVER	23
14.	REMEDIES, WAIVERS AND AMENDMENTS	23
15.	ENTIRE AGREEMENT	23
16.	COUNTERPARTS	24
17.	PARTIAL INVALIDITY	24
18.	RESERVATION OF RIGHTS AND TERMINATION	24
19.	PARTIES' RIGHTS AND OBLIGATIONS	24
20.	NOTICES	25
21.	GOVERNING LAW	26
22.	ENFORCEMENT	26
SCH	EDULE 1 OBLIGORS AND PARTICIPATING STAKEHOLDERS	27
SCH	EDULE 2 OUTSTANDINGS AND ALLOCATION TABLE	33

SCHEDULE 3 NEW SVL DOCUMENTS	42
SCHEDULE 4 NEW VESSELCO PARTY DOCUMENTS	46
SCHEDULE 5 NEW SHIPOWNER DOCUMENTS	49
SCHEDULE 6 NEW BONDHOLDER DOCUMENTS	54
SCHEDULE 7 NEW LENDER DOCUMENTS	55
SCHEDULE 8 NEW IC DOCUMENTS	57
SCHEDULE 9 NEW MILLENIUM DOCUMENTS	58
SCHEDULE 10 NEW RELATED PARTIES DOCUMENTS	59
SCHEDULE 11 NEW HHI PARTIES DOCUMENTS	61
SCHEDULE 12 ACCESSION DEED	62
SCHEDULE 13 SUBSCRIPTION LETTER	64
SCHEDULE 14 RESTRUCTURING COMPLETION LETTER	68
SCHEDULE 15 REGISTRATION RIGHTS	73
SCHEDIUE 16 INITIAL DIRECTORS	88

THIS GLOBAL RESTRUCTURING DEED (this "Deed") is made on 2014

BETWEEN:

- (1) **ZIM INTEGRATED SHIPPING SERVICES LIMITED**, a company incorporated in Israel with its registered office at 9 Andrei Sakharov St., Haifa 31016, Israel (the "Company" or "Zim");
- (2) **THE ENTITIES** listed in Part I of Schedule 1 (*Obligors and Participating Stakeholders*) hereto (the "**Obligors**");
- (3) **THE FINANCIAL INSTITUTIONS** listed in Part II of Schedule 1 (*Obligors and Participating Stakeholders*) hereto (the "**Secured Vessel Lenders**);
- (4) **THE FINANCIAL INSTITUTIONS** listed in Part III of Schedule 1 (*Obligors and Participating Stakeholders*) hereto (the "**VesselCo Parties**");
- (5) **THE ENTITIES** listed in Part IV of Schedule 1 (*Obligors and Participating Stakeholders*) hereto (the "**Shipowners**");
- (6) **THE ENTITIES** listed in Part V of Schedule 1 (*Obligors and Participating Stakeholders*) hereto on their own behalf and on behalf of the Bondholders (the "**Bond Trustees**");
- (7) **THE FINANCIAL INSTITUTIONS** listed in Part VI of Schedule 1 (*Obligors and Participating Stakeholders*) hereto (the "**Lenders**");
- (8) **ISRAEL CORPORATION LTD.**, a company incorporated in Israel with its registered office at Millennium Tower, 23 Aranha Street, Tel Aviv 61204, Israel ("**IC**");
- (9) **MILLENIUM INVESTMENTS ELAD LTD.,** a company incorporated in Israel with its registered office at 9 Andrei Sakharov St., Haifa 31016, Israel ("**Millenium**");
- (10) **THE ENTITIES** listed in Part VII of Schedule 1 (*Obligors and Participating Stakeholders*) hereto (the "**Related Parties**"); and
- (11) **THE ENTITIES** listed in Part VIII of Schedule 1 (*Obligors and Participating Stakeholders*) hereto (the "**HHI Parties**").

RECITALS

- (A) Following a period of financial difficulties of the Company, the Company, Bond Trustees, HHI Parties, IC, Lenders, Millenium, Related Parties, Secured Vessel Lenders, Shipowners, VesselCo Parties and certain other stakeholders entered into negotiations, with the objective of reaching an agreement for the financial restructuring of the Group.
- (B) The Parties have agreed the terms of a financial restructuring of the Group involving, among other things, a substantial deleveraging of the Group, issuance and allocation of Series 1 Notes and/or Series 2 Notes and ordinary shares in the Company to the Company's creditors and other Participating Stakeholders (in accordance with the Outstandings and Allocation Table), and a new equity investment by IC.

LONDON:472977.4148

(C) In order to further facilitate and to co-ordinate the implementation of the financial restructuring, the Parties have agreed to enter into this Global Restructuring Deed.

THE PARTIES AGREE AS FOLLOWS:

1. DEFINITIONS AND INTERPRETATION

- 1.1 In this Deed.
- "Accession Deed" means a document substantially in the form set out in Schedule 12 (Form of Accession Deed).
- "Additional Participating Stakeholder" means any person that becomes a Participating Stakeholder in accordance with clause 9 (*Transfers*) and clause 10 (*Accession*).
- "Affiliate" means, in relation to any person, any funds managed or advised by that person, a Subsidiary of that person or a Holding Company of that person or any other Subsidiary of that Holding Company.
- "Authorisation" means an authorisation, consent, approval, resolution, licence, exemption, filing, notarisation or registration.
- "Bondholders" means the holders of bonds under any of Zim's Series A bonds and Series B bonds (each dated November 30th 2009 as amended on July 11th 2012 and in respect of which Union Bank Trust Company Ltd. is acting as trustee) and the holders of bonds under Zim's Series C bonds (dated November 30th 2009 (as amended on July 11th, 2012) in respect of which Hermetic Trust (1975) Ltd. is acting as trustee), and "Bondholder" means any of them.
- "Bond Trustee" means each of Union Bank Trust Company Ltd., bond trustee under Zim's Series A and B bonds (dated November 30th 2009 as amended on July 11th 2012) and Hermetic Trust (1975) Ltd., bond trustee under Zim's Series C bonds (dated November 30th 2009 as amended on July 11th, 2012), and "Bond Trustees" means both of them.
- "BNPP Kexim Facility" means the loan agreement dated 4 October 2007 originally between, inter alia, Pelican Maritime (S347) Company Ltd, Pelican Maritime (S348) Company Ltd, Pelican Maritime (S393) Company Ltd, Pelican Maritime (S394) Company Ltd, and Pelican Maritime (S395) Company Ltd as borrowers, Zim as parent guarantor, BNP Paribas S.A. as facility agent, BNP Paribas S.A. as security agent and certain lenders.
- "BNPP Ksure Facility" means the loan agreement dated 26 November 2007 originally between inter alia, Flamingo Navigation (S350) Company Ltd, Flamingo Navigation (S351) Company Ltd, Pelican Maritime (S346) Company Ltd as borrowers, Zim as parent guarantor, BNP Paribas S.A. as facility agent, BNP Paribas S.A. as security agent and certain lenders.
- "Business Day" means a day on which banks are open for general business in Israel, London and New York.
- "Claims" means any liability from any actions (or omissions), causes of action, claims, judgments, executions, losses, damages, demands, suits and other liabilities (including claims in the form of debt or equity instruments) or request for reimbursement of any costs and expenses whether past, present, future, prospective or contingent, whether or not for a fixed or undetermined amount, that relates to any event or circumstance arising prior to the Restructuring Effective Time, whether or not involving payment of

money or the performance of an act or obligation, whether known or not to any party at any time, whether recognisable or unrecognisable, foreseeable or unforeseeable and however arising (whether arising at common law, in equity, by statute or pursuant to a regulation or in any other manner whatsoever) under the laws of any jurisdiction, and including any costs and expenses associated with bringing any such claim.

- "Connected Person" means, in respect of any person, such person's past, present and future, direct and indirect, Subsidiaries, shareholders, investors, funds, members, partners, and its and their respective Affiliates, officers, directors, members, partners (including, without limitation, any partnership of which such person is a general partner), any board members, employees, agents, representatives, advisors, attorneys, fiduciaries, nominees, predecessors, successors, assigns, and any other person (natural or otherwise) acting or purporting to act on behalf of any of the foregoing.
- "**Deferred Hire**" shall have the meaning given to that term in clause 6.7 (b) (*Agreement to Subscribe*).
- **Existing Debt**" means debt or overdue charter hire owing to any of the Participating Stakeholders immediately prior the Restructuring Effective Time, and which is subject to the proposed financial restructuring contemplated by this Deed, as set out in the Outstandings and Allocations Allocation Table.
- "Existing Documents" means all of the agreements, documents and instruments in relation to the Existing Debt.
- "Existing SVL Debt" means the Existing Debt owing to the Secured Vessel Lenders (or any of them), as specified in the Outstandings and Allocations Allocation Table.
- "Existing VesselCo Debt" means the Existing Debt owing to the Syndicate Lenders (or any of them), as specified in the Outstandings and Allocations Allocation Table.
- "Facility Agent" means the facility agent under any of the Syndicated Facilities.
- "Group" means the Company and each of its Subsidiaries from time to time.
- "Holding Company" means, in relation to a company or corporation, any other company or corporation in respect of which it is a Subsidiary.
- **"HHI Subordinated Loan Agreement"** means the loan agreement between Hyundai Samho Heavy Industries Co. Ltd. and Zim dated on or about the Restructuring Effective Time.
- "IC Investment Agreement" means the IC investment agreement referred to under the heading of "Israel Corporation" in Schedule 8 (New IC Documents).
- "Initial Directors" means (i) the directors listed in paragraph 1 in Schedule 16 (*Initial Directors*) which includes the identity of the first board of directors, and (ii) the additional initial directors to be nominated to the Board of Directors of the Company in accordance with the nomination process detailed in paragraph 2 in Schedule 16 (*Initial Directors*).
- "June 2014 Business Plan" means the business plan so entitled provided to the Participating Stakeholders in the VDR.

"Legal Reservations" means:

- (a) the principle that equitable remedies may be granted or refused at the discretion of a court and the limitation of enforcement by laws relating to insolvency, reorganisation and other laws generally affecting the rights of creditors;
- (b) the time barring of claims under the Limitation Acts, the possibility that an undertaking to assume liability for or indemnify a person against non-payment of UK stamp duty may be void and defences of set-off or counterclaim;
- (c) similar principles, rights and defences under the laws of any relevant jurisdiction; and
- (d) any other matters which are set out as qualifications or reservations as to matters of law of general application in case of any legal opinions issued in connection with the New Documents or, if no legal opinions are given, would be customarily included in legal opinions.

"Limitation Acts" means the Limitation Act 1980 and the Foreign Limitation Periods Act 1984.

"Long-Stop Time" means 23:59 hours Israel time on 310 July 2014.

"New Bondholder Documents" has the meaning given to that term in clause 3.11.

"New Documents" means all New Bondholder Documents, New HHI Documents, New IC Documents, New Millenium Documents, New Lender Documents, New Related Parties Documents, New Shipowner Documents, New SVL Documents and New VesselCo Party Documents.

"New HHI Documents" has the meaning given to that term in clause 3.14.

"New IC Documents" has the meaning given to that term in clause 3.20.

"New Lender Documents" has the meaning given to that term in clause 3.14.

"New Millenium Documents" has the meaning given to that term in clause 3.18.

"New Related Parties Documents" has the meaning given to that term in clause 3.19.

"New Shipowner Documents" has the meaning given to that term in clause 3.9.

"New SVL Documents" has the meaning given to that term in clause 3.2.

"New VesselCo Party Documents" has the meaning given to that term in clause 3.6.

"Outstandings and Allocation Table" means the table set out in Schedule 2 (Outstandings and Allocation Table).

"Participating Stakeholder" means each of the entities listed in Schedule 1 (Obligors and Participating Stakeholders) hereto, other than the Obligors.

"Qualifying Investor" means any investor complying with either section 15A(b)(1) or 15A(b)(2) of the Israel Securities Law of 1968, as shall be updated from time to time.

- "Party" means a party to this Deed.
- "Registration Rights Schedule" means the schedule attached as Schedule 15.
- "Related Funds" in relation to a fund (the "first fund") means a separate fund which is managed or advised by the same investment manager or investment adviser as the first fund or, if it is managed by a different investment manager or investment adviser, a separate fund whose investment manager is an Affiliate of the investment manager or investment adviser of the first fund.
- "Restructuring" means the restructuring of the Company's capital structure as set out in the Outstandings and Allocation Table and as contemplated by the New Documents.
- "Restructuring Completion Letter" means the letter substantially in the form attached as Schedule 14 (Restructuring Completion Letter).
- "Restructuring Effective Time" means the time at which Zim executes and delivers to all other Parties the Restructuring Completion Letter.
- "Security Agent" means the security agent under any of the Syndicated Facilities.
- "Series 1 Notes" mean the notes constituted by a trust indenture entered into by the Company and Hermetic Trust (1975) Ltd. as trustee, dated on or about the date hereof.
- "Series 2 Notes" mean the notes constituted by a trust indenture entered into by the Company and Hermetic Trust (1975) Ltd. as trustee, dated on or about the date hereof.
- "Settlement Instructions" means (in respect of the Series 1 Notes or Series 2 Notes) each of (i) the name of the security account with a member of the TASE to which the notes will be settled, the TASE member name, the TASE member number, the branch name and account number of the entity to hold the Series 1 Notes or Series 2 Notes (as relevant); (ii) the full legal name of the entity to hold the Series 1 Notes or Series 2 Notes, its incorporation number (if any), jurisdiction of incorporation, mailing address and e-mail address; and (in respect of equity) the full legal name of the entity to hold the equity, its incorporation number (if any), jurisdiction of incorporation, mailing address and e-mail address.
- "Subscription Letter" means a letter substantially in the form set out in Schedule 13 (Subscription Letter) to be entered into by certain Participating Stakeholders other than IC.
- "Subsidiary" means in relation to any company, corporation or other legal entity, (a "holding company"), a company, corporation or other legal entity:
- (a) which is controlled, directly or indirectly, by the holding company;
- (b) more than half the issued share capital of which is beneficially owned, directly or indirectly, by the holding company; or
- (c) which is a subsidiary of another Subsidiary of the holding company,

and, for this purpose, a company or corporation shall be treated as being controlled by another if that other company or corporation is able to determine the composition of the majority of its board of directors or equivalent body.

"Syndicate Lender" means a lender under any of the Syndicated Facilities.

- "Syndicated Facilities" means the BNPP Kexim Facility, BNPP Ksure Facility, Wilmington 345 Facility, Wilmington 349 Facility and Wilmington 352 Facility, or any of them.
- "TASE" means the Tel Aviv Stock Exchange.
- "Third Parties Act" means the Contracts (Rights of Third Parties) Act 1999.
- "VDR" means the Company's virtual data room.
- "Wilmington 345 Facility" means the loan agreement dated 5 February 2007 originally between, inter alia, Pelican Maritime (S345) Company Ltd. as borrower, Zim as guarantor and KfW IPEX-Bank GmbH as facility agent and certain lenders.
- "Wilmington 349 Facility" means the loan agreement dated 5 February 2007 originally between, inter alia, Flamingo Navigation (S349) Company Ltd. as borrower, Zim as guarantor and KfW IPEX-Bank GmbH as facility agent and certain lenders.
- "Wilmington 352 Facility" means the loan agreement dated 5 February 2007 originally between, inter alia, Flamingo Navigation (S352) Company Ltd. as borrower, Zim as guarantor and KfW IPEX-Bank GmbH as facility agent and certain lenders.
- 1.2 In this Deed unless a contrary indication appears:
 - (a) reference to any agreement or instrument is a reference to that agreement or instrument as amended, novated, supplemented, extended, restated or replaced;
 - (b) references to any party herein shall be construed so as to include that party's respective successors in title, permitted assignees and permitted transferees;
 - (c) one gender includes all genders, and references to the singular includes the plural and vice versa;
 - (d) a **person** includes any individual, company, corporation, unincorporated association or body (including a partnership, trust, joint venture or consortium), government, state, agency, organisation or other entity whether or not having separate legal personality and their successors in title, assigns and transferees permitted pursuant to the applicable agreement or instrument;
 - (e) references to clauses, paragraphs and Schedules shall be construed as references to clauses and paragraphs of, and Schedules to, this Deed;
 - (f) headings in this Deed are inserted for convenience and do not affect its interpretation;
 - (g) **include** and **including** shall be construed without limitation; and
 - (h) references to this Deed include its Schedules.
- 1.3 It is intended that this Deed takes effect as a deed notwithstanding the fact that a party may only execute this document under hand.
- 1.4 In this Deed, any undertaking, obligation or commitment provided or assumed by a Bond Trustee is assumed and provided by the Bond Trustee on behalf of each of the Bondholders under the

bond trust deed relating to the indebtedness incurred by Zim in respect of which that Bond Trustee is acting as a trustee, unless expressly stated otherwise, and any rights or benefits provided to a Bond Trustee are assumed by and provided to the Bond Trustee on behalf of itself, each member of the Bondholders' committee and each of the Bondholders under the bond trust deeds relating to indebtedness incurred by Zim in respect of which that Bond Trustee is acting as trustee.

1.5 References to the 'articles of association of the Company in the form agreed by the Parties' shall be to the form of the articles last posted in the Company's VDR on or prior to the Restructuring Effective Time.

2. EFFECTIVENESS OF THIS DEED AND LONG-STOP TIME

This Deed is effective on the date it is signed by all Parties. The Company agrees to notify the Participating Stakeholders in writing promptly once all Parties have executed this Deed. If the Restructuring Effective Time has not occurred by the Long-Stop Time, this Deed will terminate automatically and be of no further force and effect.

3. CONFIRMATIONS

3.1 Unless otherwise stated, all confirmations given in this clause 3 are given on each of (i) the date of this AgreementDeed; and (ii) both immediately before and at the Restructuring Effective Time, in each case by reference to the facts and circumstances then existing on such date or at such time, as applicable.

Secured Vessel Lenders (Excluding VesselCo Parties)

3.2 Outstandings

Each Secured Vessel Lender confirms to each other Party that:

- (a) all documents listed under its name in Schedule 3 (*New SVL Documents*) (such Secured Vessel Lender's "**New SVL Documents**"), together with its allocation of Series 1 Notes and equity (as relevant) in each case as set out in the Outstandings and Allocation Table, comprise all of the arrangements, agreements and understandings that will be in place between it and the Group as at the Restructuring Effective Time;
- (b) the New SVL Documents will be in full force and effect upon satisfaction of the conditions precedent described therein (including the issuance by Zim of the Restructuring Completion Letter);
- (c) the outstandings listed opposite its name in the Outstandings and Allocation Table (in each case where it appears, as relevant) represent in aggregate all amounts due to it from the Group on the dates specified therein; and
- (d) it has executed all New SVL Documents to which it is expressed to be a party.

3.3 Execution by Secured Vessel Lenders

Each of the Secured Vessel Lenders is entering into this Deed in its capacity as a lender and only in respect of the Existing SVL Debt which it holds and not in any other capacity.

3.4 Nomination of Initial Directors and approval of articles of association

With effect from the Restructuring Effective Time, each of the Secured Vessel Lenders (which has subscribed for shares in the Company) approves, in its capacity as a shareholder of the Company and for and on behalf of any Designated Recipient, the nomination of the Initial Directors and the process for nominating additional Initial Directors—set out in Schedule 16 (Initial Directors) and the amended articles of association of the Company in the form agreed by the Parties as at the Restructuring Effective Time.

VesselCo Parties

3.5 Lenders of Record

Each Facility Agent confirms to each other Party that as at the date of this Deed each lender under the Syndicated Facility in respect of which it is acting as facility agent is listed as a Party to this Deed as a Syndicate Lender under the heading of that Syndicated Facility in Part III of Schedule 1 (Obligors and Participating Stakeholders).

3.6 Outstandings

Each Syndicate Lender confirms to each other Party (in respect of itself only) that:

- all documents listed in Schedule 4 (*New VesselCo Party Documents*) under the heading of the facility in respect of which it is indicated as being a Syndicate Lender (such Party's "New VesselCo Party Documents"), together with its allocation of Series 1 Notes and equity (as relevant) in each case as set out in the Outstandings and Allocation Table, comprise all of the arrangements, agreements and understandings that will be in place between it and the Group as at the Restructuring Effective Time and its commitments and outstandings under that facility and its position as lender of record under that facility have not changed since the date of this Deed;
- (b) the New VesselCo Party documents will be in full force and effect upon satisfaction of the conditions precedent described therein (including the issuance by Zim of the Restructuring Completion Letter);
- (c) the outstandings listed opposite its name in the Outstandings and Allocation Table (in each case where it appears, as relevant) represent in aggregate all amounts due to it from the Group on the dates specified therein; and
- (d) it has executed all New VesselCo Party Documents to which it is expressed to be a party.

3.7 Execution by VesselCo Parties

Each Syndicate Lender is entering into this Deed in its capacity as a lender and only in respect of the Existing VesselCo Debt which it holds and not in any other capacity. Each Facility Agent is entering into this Deed in its capacity as such in respect of the Syndicated Facility for which it is acting in such capacity and not in any other capacity and only for the purpose of providing the confirmation given in clause 3.5 (*Lenders of Record*).

3.8 Nomination of Initial Directors and approval of articles of association

With effect from the Restructuring Effective Time, each Syndicate Lender (which has subscribed for shares in the Company) approves, in its capacity as a shareholder of the Company and for and on behalf of any Designated Recipient, the nomination of the Initial Directors and the process for nominating additional Initial Directors set out in Schedule 16 (*Initial Directors*) and the amended articles of association of the Company in the form agreed by the Parties as at the Restructuring Effective Time.

Shipowners

- 3.9 Each Shipowner confirms to each other Party that:
 - (a) all documents listed under its name in Schedule 5 (*New Shipowner Documents*) (such Shipowner's "**New Shipowner Documents**"), together with its allocation of Series 1 Notes, Series 2 Notes and equity (as relevant) in each case as set out in the Outstandings and Allocation Table, comprise all of the arrangements, agreements and understandings that will be in place between it and the Group as at the Restructuring Effective Time;
 - (b) the New Shipowner Documents will be in full force and effect upon satisfaction of the conditions precedent (if any) described therein (including the issuance by Zim of the Restructuring Completion Letter);
 - (c) the outstandings listed opposite its name in the Outstandings and Allocation Table (in each case where it appears, as relevant) represent in aggregate all amounts due to it from the Group on the dates specified therein (disregarding current amounts due under current charter arrangements (taking into account amendments effective as at the Restructuring Effective Time) and/or ship operation agreements between it and Zim as part of ordinary course arrangements on arm's length terms under those charters (including for example current charter hire accrued or due for payment but unpaid or reimbursement for ordinary course expenses such as, for example fuel charges);
 - (d) it has executed all New Shipowner Documents to which it is expressed to be a party; and
 - (e) as at the Restructuring Effective Time, no Shipowner is aware of any Claim (other than (i) in respect of Deferred Hire, (ii) as set out in clause 7.2(a)(ii)(C) or (iii) in respect of on-going ordinary course P&I claims of an operational nature involving Zim and vessels owned or operated by the Shipowners which (in each case) do not involve the payment of hire or accrued or deferred hire or callable exchange notes) that could be made against any member of the Group under an existing charter party arrangement.
- 3.10 Nomination of Initial Directors and approval of articles of association

With effect from the Restructuring Effective Time, each Shipowner (which has subscribed for shares in the Company) approves, in its capacity as a shareholder of the Company and for and on behalf of any Designated Recipient, the nomination of the Initial Directors and the process for nominating additional Initial Directors set out in Schedule 16 (*Initial Directors*) and the amended

articles of association of the Company in the form agreed by the Parties as at the Restructuring Effective Time.

Bondholders

- 3.11 Each Bond Trustee confirms, on its own behalf and on behalf of the Bondholders in respect of which it is acting as trustee, to each other Party that:
 - (a) all documents listed under its name in Schedule 6 (*New Bondholder Documents*) (such Bond Trustee's "**New Bondholder Documents**"), together with its allocation of Series 1 Notes and equity (as relevant) in each case as set out in the Outstandings and Allocation Table, comprise all of the arrangements, agreements and understandings that will be in place as at the Restructuring Effective Time between the Group and the Bondholders (in their capacity as such) under the bond series in respect of which the Bond Trustee is acting as trustee, and those documents will be in full force and effect upon satisfaction of the conditions precedent described therein (including the issuance by Zim of the Restructuring Completion Letter);
 - (b) the outstandings listed opposite the references "Bondholders Union Bank Trust Company Ltd. (Series A)", "Bondholders Union Bank Trust Company Ltd. (Series B)" and "Bondholders Hermetic Trust (1975) Ltd. (Series C)" in the Outstandings and Allocation Table represent all amounts due to the Bondholders from the Group on the dates specified therein; and
 - (c) under the New Bondholder Documents the amounts set out in paragraph (b) above will be released in full as part of the cancellation of the bonds and in consideration for the issuance of Series 1 Notes and equity as set out in the Outstandings and Allocation Table.
- 3.12 Hermetic Trust (1975) Ltd. confirms that it has executed all New Bondholder Documents to which it is expressed to be a party.
- 3.13 Nomination of Initial Directors and approval of articles of association

With effect from the Restructuring Effective Time each Bond Trustee, on its own behalf and on behalf of the Bondholders in respect of which it is acting as trustee, in their capacity as shareholders of the Company, approves the nomination of the Initial Directors and the process for nominating additional Initial Directors set out in Schedule 16 (*Initial Directors*) and the amended articles of association in the form agreed by the Parties as at the Restructuring Effective timeTime.

Lenders (excluding Secured Vessel Lenders and VesselCo Parties)

- 3.14 Each Lender confirms to each other Party that:
 - (a) all documents listed under its name in Schedule 7 (*New Lender Documents*) (such Lender's "**New Lender Documents**"), together with its allocation of Series 1 Notes and equity (as relevant) in each case as set out in the Outstandings and Allocation Table,

- comprise all of the arrangements, agreements and understandings that will be in place between it and the Group as at the Restructuring Effective Time;
- (b) the New Lender Documents will be in full force and effect upon satisfaction of the conditions precedent described therein (including the issuance by Zim of the Restructuring Completion Letter);
- (c) the outstandings listed opposite its name in the Outstandings and Allocation Table (in each case where it appears, as relevant) represent in aggregate all amounts due to it from the Group on the dates specified therein; and
- (d) it has executed all New Lender Documents to which it is expressed to be a party.

3.15 Nomination of Initial Directors and approval of articles of association

With effect from the Restructuring Effective Time, each of the Lenders (which has subscribed for shares in the Company) approves, in its capacity as a shareholder of the Company and for and on behalf of any Designated Recipient, the nomination of the Initial Directors and the process for nominating additional Initial Directors set out in Schedule 16 (*Initial Directors*) and the amended articles of association of the Company in the form agreed by the Parties as at the Restructuring Effective Time.

IC

3.16 IC confirms to each other Party that:

- (a) all documents listed under its name in Schedule 8 (*New IC Documents*) (the "**New IC Documents**"), comprise all of the arrangements, agreements and understandings that will be in place between it and the Group as at the Restructuring Effective Time, and those documents will be in full force and effect upon satisfaction of the conditions precedent described therein (including the issuance by Zim of the Restructuring Completion Letter);
- (b) the outstandings listed opposite its name in the Outstandings and Allocation Table represent in aggregate all amounts due to it from the Group on the dates specified therein;
- (c) it has executed all New IC Documents to which it is expressed to be a party; and
- (d) immediately following the Restructuring Effective Time no amount, liability or obligation will be due, outstanding or accruing by any member of the Group (whether alone or jointly, or jointly and severally, with any other person, whether actually or contingently, and whether as principal, surety or otherwise) to IC other than under the New IC Documents.

3.17 Nomination of Initial Directors and approval of the articles of association

With effect from the Restructuring Effective Time, IC hereby approves and confirms to each other Party (in its capacity as shareholder of the Company) the nomination of the Initial Directors and the process for nominating additional Initial Directors set out in Schedule 16 (Initial Directors) and the amended articles of association of the Company in the form agreed by the Parties as at the Restructuring Effective Time.

Millenium

- 3.18 Millenium confirms to each other Party that:
 - (a) all documents listed under its name in Schedule 9 (*New Millenium Documents*) (the "**New Millenium Documents**"), comprise all of the arrangements, agreements and understandings that will be in place between it and the Group as at the Restructuring Effective Time, and those documents will be in full force and effect upon satisfaction of the conditions precedent described therein (including the issuance by Zim of the Restructuring Completion Letter);
 - (b) the outstandings listed opposite its name in the Outstandings and Allocation Table represent in aggregate all amounts due to it from the Group on the dates specified therein;
 - (c) it has executed all New Millenium Documents to which it is expressed to be a party; and
 - (d) immediately following the Restructuring Effective Time no amount, liability or obligation will be due, outstanding or accruing by any member of the Group (whether alone or jointly, or jointly and severally, with any other person, whether actually or contingently, and whether as principal, surety or otherwise) to Millenium.

Related Parties

- 3.19 Each Related Party confirms to each other Party that:
 - (a) all documents listed under its name in Schedule 10 (New Related Parties Documents) (such Related Party's "New Related Parties Documents") comprise all of the arrangements, agreements and understandings that will be in place between it and the Group as at the Restructuring Effective Time (other than commercial transactions on arm's length terms having an aggregate transaction value across the Group of less than \$5,000,000) and those documents will be in full force and effect upon satisfaction of the conditions precedent described therein (including the issuance by Zim of the Restructuring Completion Letter);
 - (b) the outstandings listed opposite its name in the Outstandings and Allocation Table represent in aggregate all amounts due to it from the Group on the dates specified therein (disregarding current amounts due under current charter arrangements (taking into account amendments effective at the Restructuring Effective Time) and/or ship operation agreements between it and Zim as part of ordinary course arrangements on arm's length terms under those charters (including for example current charter hire accrued or due for payment but unpaid or reimbursement for ordinary course expenses such as, for example fuel charges));
 - (c) it has executed all New Related Parties Documents to which it is expressed to be a Party;
 - (d) immediately following the Restructuring Effective Time no amount, liability or obligation will be due outstanding or accruing by any member of the Group (whether alone or jointly, or jointly and severally, with any other person, whether actually or contingently, and whether as principal, surety or otherwise) to any Related Party other than under the New Related Parties Documents (other than amounts, liabilities or obligations due outstanding or accruing under commercial transactions between the

Group and the Related Parties on arm's length terms having an aggregate transaction value across the Group of less than \$5,000,000 and disregarding current amounts due under current charter arrangements and/or ship operation agreements between it and Zim as part of ordinary course arrangements under those charters (including for example current charter hire accrued or due for payment but unpaid or reimbursement for ordinary course expenses such as, for example fuel charges)); and

(e) as at the Restructuring Effective Time, no Related Party is aware of any Claim (other than in respect of (i) Deferred Hire or (ii) on-going ordinary course P&I claims of an operational nature involving Zim and vessels owned or operated by the Related Parties which (in each case) do not involve the payment of hire or accrued or deferred hire or callable exchange notes) that could be made against any member of the Group under an existing charter party arrangement.

HHI Parties

- 3.20 Each HHI Party confirms to each other Party that:
 - all documents listed under its name in Schedule 11 (*New HHI Documents*) (the "New HHI Documents"), together with its allocation of Series 1 Notes and the HHI Subordinated Loan Agreement (as relevant) in each case as set out in the Outstandings and Allocation Table, comprise all of the arrangements, agreements and understandings that will be in place as at the Restructuring Effective Time between the Group and the HHI Parties, and those documents will be in full force and effect upon satisfaction of the conditions precedent described therein (including the issuance by Zim of the Restructuring Completion Letter);
 - (b) the outstandings listed opposite its name in the Outstandings and Allocation Table represent all amounts due to it from the Group on the dates specified therein; and
 - (c) it has executed all New HHI Documents to which it is expressed to be a party.

All Participating Stakeholders

3.21 All Participating Stakeholders confirm that they have not charged or received any amount by way of default interest, penalties or similar late charges in respect of amounts owing to them which are the subject of the transactions the subject of this Deed.

4. REPRESENTATIONS

4.1 Representations from all Parties

Each Party (other than the Bond Trustees and the Facility Agents) makes the representations and warranties set out in sub-paragraphs (a) to (e) in this clause 4.1 to each other Party (including the Bond Trustees and the Facility Agents). Such representations and warranties are made or deemed to be made at the times and in the manner set out at clause 4.5 (*Times when representations made*):

(a) it is duly incorporated (if a corporate person) or duly established (in any other case) and validly existing under the laws of its jurisdiction of incorporation or formation.

- (b) the obligations expressed to be assumed by it in this Deed and each New Document to which it is, or will be, a party are legal, valid, binding and enforceable on it, subject to any applicable Legal Reservations;
- (c) the entry into and performance by it of, and the transactions contemplated by, this Deed and each New Document to which it is, or will be, a party do not, and will not, conflict with any law or regulation applicable to it or with any of its articles of association, memorandum of association or any other constitutional documents;
- (d) it has the power to enter into, perform and deliver, and has taken all necessary action to authorise its entry into, performance and delivery of this Deed, each New Document to which it is, or will be, a party and (subject to fulfilment or waiver of any conditions precedent to any obligations contained therein) the transactions contemplated by this Deed and such New Documents; and
- (e) all necessary Authorisations required (i) for the performance by it of this Deed and each New Document to which it is, or will be, a party and the transactions contemplated by this Deed and such New Documents and (ii) to make this Deed and each New Document to which it is, or will be, a party admissible in evidence in its jurisdiction of incorporation have been obtained or effected and are in full force and effect.
- 4.2 Each Bond Trustee makes the representations and warranties set out in paragraphs (a) to (db) below (in respect of itself only). Such representations and warranties are made or deemed to be made at the times and in the manner set out at clause 4.5 (*Times when representations made*):
 - (a) it is a company registered in Israel and engaged in trusts; and
 - (b) the Bondholders' meetings voted on, inter alia, the approval of the Restructuring and the authorisation of the Bond Trustees by each of the Series A, Series B and Series C bondholders (as relevant, in each case representing the bond series for which it is acting as trustee) to execute this Deed on behalf of the relevant series of bonds.
- 4.3 Additional representations of Zim and the Obligors

Zim and (where indicated below) each Obligor in its own capacity (and solely in respect of itself) makes the representations and warranties set out in this clause 4.3 to each other Party. Such representations and warranties are made or deemed to be made at the times and in the manner set out in clause 4.5 (*Times when representations made*):

- (a) it is not the legal owner of, nor does it have any beneficial interest in, any Existing Debt;
- (b) to the best of its knowledge, having made all reasonable enquiries, no order has been made, petition presented or resolution passed for the winding-up of or appointment of a liquidator, administrative receiver, administrator, compulsory manager, trustee, custodian, sequestrator or other similar officer in respect of it or (in the case of Zim only) any other member of the Group and no analogous procedure has been commenced in any jurisdiction which has not been discharged or resolved or otherwise lapsed;
- (c) it is not a "Company in Violation" under section 362A of the Israeli Companies Law 5759-1999;

- (d) it has full power and authority to own its property and assets and to carry on its business as it is now being conducted;
- (e) the choice of governing law of this Deed and each New Document to which it is a party will be recognised and enforced in its jurisdiction of incorporation, subject to any applicable Legal Reservations; and
- (f) it will not be entitled to claim immunity from suit or other legal process in any proceedings taken in its jurisdiction of incorporation in relation to this Deed or any New Document.

4.4 Additional representations of Zim:

- (a) Any factual information relating to the Group comprised in the June 2014 Business Plan is, taken as a whole, true and accurate in all material respects as at the date it is provided to the Participating Stakeholders or (if different) as at the date at which it is stated to apply.
- (b) Nothing has occurred or been omitted from and no information has been given or withheld that results in any of the information relating to the Group comprised in the June 2014 Business Plan, taken as a whole, being untrue or misleading in any material respect as at the date it is provided to the Participating Stakeholders or (if different) as at the date at which it is stated to apply.
- (c) Any financial projections contained in the June 2014 Business Plan have been prepared in good faith on the basis of recent information and on the basis of assumptions believed by Zim to be reasonable at the time they were provided, it being understood that projections may be subject to significant market uncertainties or third party actions, which are beyond the control of the Group.
- (d) All New Documents have been executed by the members of the Group who are parties to those documents.
- (e) With effect from the Restructuring Effective Time, the documents set out in Schedules 3 through 11 (inclusive) set out all the arrangements, agreements and understandings with the counterparties referred to in those schedules (other than with the Related Parties).
- (f) With effect from the Restructuring Effective Time, all arrangements, agreements and understandings between the Group and the Related Parties are either set out in Schedule 10 (*New Related Parties Documents*) or are commercial transactions on arm's length terms having an aggregate transaction value across the Group of less than \$5,000,000.
- (g) As at the Restructuring Effective Time (other than (i) in respect of Deferred Hire, (ii) as set out in clause 7.2(a)(ii)(C) or (iii) in respect of on-going ordinary course P&I claims of an operational nature involving Zim and vessels owned or operated by the Shipowners or Related Parties which (in each case) do not involve the payment of hire or accrued or deferred hire or callable exchange notes), neither Zim nor any Obligor is aware of any Claim that (y) could be made against a Related Party or a Shipowner under an existing charter party arrangement or (z) could be brought by a Related Party or Shipowner under an existing charter party arrangement.

(h) The new equity, when issued and allotted in accordance with this Deed, will be duly authorised, validly issued, fully paid, and non-assessable, and upon issuance, free of any preemption rights, will have the rights, preferences, privileges and restrictions set forth in the articles of the Company, and will be free and clear of any liens, claims, encumbrances or third party rights of any kind and duly registered in the names of the relevant Participating Stakeholders (or Designated Recipient (as defined in Clause 6.4) as relevant) in accordance with the Outstandings and Allocations Allocation Table in the Company's share register and will be offered, sold and issued in compliance with all applicable securities laws.

4.5 Times when representations made

The representations and warranties in this clause 4 are made by each relevant Party on each of the following specified dates or upon the occurrence of the following specified times, as applicable:

- (a) the date of this Deed; and
- (b) immediately before and at the Restructuring Effective Time,

in each case by reference to the facts and circumstances then existing on any such date or at such time, as applicable.

5. RELATIONSHIP WITH OTHER DOCUMENTS

- 5.1 Until the Restructuring Effective Time, the Existing Documents shall continue in full force and effect and this Deed shall be without prejudice to the Parties' respective rights under the Existing Documents (or otherwise). The relevant Parties to the Existing Documents shall continue to comply with the terms of all such Existing Documents until such time as the Existing Documents are no longer effective pursuant to the terms of this Deed and/or the New Documents (as applicable).
- 5.2 Each relevant Party fully reserves any and all of its rights under the Existing Documents to which it is a party or this Deed to the extent not expressly amended, varied or waived by this Deed or the New Documents to which it is a party.

6. AGREEMENT TO SUBSCRIBE

- 6.1 The Company agrees to offer to each Party whose name is listed in the Outstandings and Allocation Table an allocation of Series 1 Notes, Series 2 Notes and/or equity (as relevant) equal to those set out against the name of the relevant Parties in the Outstandings and Allocation Table and subject to receipt by the Company of applicable Settlement Instructions and the Subscription Letter (together "Subscription Notices").
- 6.2 Subject only to its obligations under section 2 of Schedule 15 (Registration Rights), nothing in this Deed shall oblige the Company to publish a prospectus. The Company confirms that as at the Restructuring Effective Time there were not more than 15 Parties or Designated Recipients subscribing for Series 1 Notes, Series 2 Notes and/or equity who were not Qualifying Investors.

 The Until the date which is six months after the Restructuring Effective Time, the Company will, within 3 Business Days of a request by any other Party, confirm the number of Parties or Designated Recipients currently holding Series 1 Notes, Series 2 Notes and/or equity who are not Qualifying Investors.

- A Party can accept (or designate in accordance with clause 6.4 below) its allocation of each class of instrument or any part thereof in one or more Subscription Notices. No Party shall have any liability for failure to deliver Subscription Notices, but no equity of Zim or Series 2 Notes or (except in the case of Bondholders) Series 1 Notes will be issued to any Party unless that Party has delivered duly completed Subscription Notices to Zim within six months of the Restructuring Effective Time. Notwithstanding the other provisions of this clause 6, no Party will have any right to be allotted or issued any equity or, Series 2 Notes or (except in the case of Bondholders) Series 1 Notes if that Party or its Designated Recipient (as defined below) has not provided the applicable Subscription Notices within such six month period. Each Party agrees that it will not be entitled to, or entitled to designate, any allocation of Series 1 Notes, Series 2 Notes and/or equity of Zim except to the extent of the allocation set out opposite its name in the Outstandings and Allocation Table.
- Any Party (such Party, a "Designating Party") may, before being issued Series 1 Notes, Series 2 6.4 Notes and/or equity, elect to designate another entity (such entity, a "Designated Recipient") to own, hold or otherwise be issued with all (but not) of the Series 1 Notes, and/or all (but notor part) of the Series 2 Notes and/or all or part of the equity in Zim (in each case) to which such Designating Party is entitled as set out opposite such Designating Party's name in the Outstandings and Allocation Table. Any Designating Party intending to make this election shall notify Zim in writing of its election as soon as reasonably practicable, and upon receipt by Zim of the relevant Subscription Letter from such Designating Party, if required, and the Subscription Notices from such Designated Recipient (in each case) within six months of the Restructuring Effective Time (and so long as such Designated Recipient is otherwise entitled to hold Series 1 Notes, Series 2 Notes and/or equity (as relevant)), Zim shall allocate such Series 1 Notes, Series 2 Notes and/or equity the subject of such election to such Designated Recipient. During such sixmonth period after the Restructuring Effective Time and pending such allocation, the Party shall remain entitled to be issued Series 1 Notes, Series 2 Notes and equity (as relevant) to the extent set out opposite such Party's name in the Outstandings and Allocation Table.
- 6.5 With effect from the Restructuring Effective Time, each Party whose name is listed in the Outstandings and Allocation Table (other than IC, whose subscription is as described below) has the right to subscribe (whether for itself or by designation as set out in clause 6.4) for the Series 1 Notes, Series 2 Notes and/or equity of Zim (as relevant) in each case to the extent (if any) set out opposite its name in the Outstandings and Allocation Table and for a total consideration equal to the aggregate amount set out opposite its name in that table (in each place it appears, where relevant) under the heading "Total deficiency claim immediately prior to Restructuring Effective Time".
- 6.6 In consideration for the Company's offer to allot Series 1 Notes, Series 2 Notes and/or equity and in full satisfaction of the consideration described in clause 6.5, each Participating Stakeholder has agreed to execute any releases set out in the New Documents to which it is expressed to be a Party.
- 6.7 In consideration for the Company's offer to allot Series 2 Notes and/or equity to the Shipowners and in full satisfaction of the consideration in respect of the Shipowners' allocations described in clause 6.5 and the other transactions contemplated in the New Shipowner Documents and New Related Parties Documents, with effect from the Restructuring Effective Time, each Shipowner and each Related Party hereby:
 - (a) irrevocably, fully and finally waives, releases and discharges the Company from all undertakings, liabilities and obligations, whether actual or contingent, direct or indirect,

- and whether past, present or future, incurred or owing (including accrued and/or capitalised interest) pursuant to the callable exchange notes issued by the Company to it (if any) pursuant to the 2009 restructuring arrangements of the Company, with the effect that such callable exchange notes shall have no further force or effect on and from the Restructuring Effective Time and the Company shall have no liability thereunder on and from the Restructuring Effective Time; and
- (b) hereby irrevocably, fully and finally (i) waives, releases and discharges the Company from any claims, liabilities and obligations relating to any hire outstanding and/or hire reductions (and/or any capitalised interest on any of the foregoing) due from or payable by the Company to it in connection with the period ending at the Restructuring Effective Time, including any such outstanding hire and/or hire reductions due or payable under any addendum signed and/or agreed upon between the Company and any Shipowner or, as the case may be, Related Party during 2013 (all such hire outstanding and/or hire reductions, the "Deferred Hire") and/or any other agreement or addendum in relation to the Deferred Hire executed between the Company and any Shipowner or Related Party which has taken effect prior to the Restructuring Effective Time, to the extent such addendum or agreement has not been already performed and (ii) waives any termination right under the relevant charterparty between it and the Company arising from the Deferred Hire. For the avoidance of doubt the following shall not be the subject of this release: (i) hire or amounts due and payable under any addenda to any charter party between the Company and a Shipowner or, as the case may be, Related Party which takes effect at the Restructuring Effective Time; and (ii) current amounts due under current charter arrangements (taking into account amendments effective at the Restructuring Effective Time) and/or ship operation agreements between it and the Company as part of ordinary course arrangements under those charters (including for example current charter hire accrued or due for payment but unpaid or reimbursement for ordinary course expenses such as, for example fuel charges).
- 6.8 IC's subscription for shares in the Company will be made under the IC Subscription Agreement.
- 6.9 The Company shall issue the duly written up physical share certificates to the Participating Stakeholders who have provided the Subscription Notices in respect of the ordinary shares to Zim as required pursuant to clause 6.1 (or Designated Recipients to the extent the provisions of clause 6.4 have been complied with) within 21 days of the day on which the Restructuring Effective Time occurs (or if later, within 21 days of the time they have provided such Subscription Notices so long as such Subscription Notices have been provided within the six month limit prescribed by clause 6.1).
- 6.10 Each Participating Stakeholder who has been issued equity shall have the rights conferred on holders of the Company's equity set out in the Registration Rights Schedule. At the request of any registered holder of the Company's equity, the Company will enter into an agreement with any new holder of equity in the Company in substantially the same form as the Registration Rights Schedule (which agreement shall require the Company to enter into any new agreement with any subsequent equity holder).
- 6.11 Any fractional entitlement to shares arising from the allocations set out in Schedule 2 (*Outstandings and Allocation Table*) shall be rounded up or down to the nearest whole number of shares as appropriate, with fractional entitlements of 0.5 or more being rounded up and fractional entitlements of less than 0.5 being rounded down.

7. MUTUAL RELEASES

- 7.1 With effect from the Restructuring Effective Time:
 - (a) (subject to clause 7.2) each Party (other than the Bond Trustees, the Bondholders and the Facility Agents) hereby irrevocably, expressly and unconditionally:
 - (i) waives, releases, acquits and discharges and agrees to ensure (to the extent capable of ensuring) that each of its respective Connected Persons waives, releases, acquits and discharges, to the fullest extent permissible by applicable law, any and all Claims (subject to any Claims arising by reason of fraud or deceit) that it ever had, now has or may have against any other Party (including any Connected Person of such other Party, and including for the avoidance of doubt the Bond Trustees and the Bondholders) (each a "Released Person"); and
 - (ii) agrees (subject to any Claims arising by reason of fraud or deceit), not to take or pursue any legal or other action (and to ensure, to the extent capable of ensuring, that none of its respective Connected Persons takes or pursues any legal or other action) in respect of any Claims against any Released Person,

(in each case) upon, in connection with, or by reason of any matter, act, omission, failure to act, transaction, event, circumstance, occurrence, cause or thing whatsoever prior to the Restructuring Effective Time (other than by reason of fraud or deceit), arising directly or indirectly in connection with the Group (including with respect to any Released Person's or any of its Connected Person's control, management or operation of the Group), the Restructuring, any of the documents implementing the Restructuring (including, without limitation, any documents relating to or the allocation of Series 1 Notes, Series 2 Notes, equity or any New Documents) and/or all other agreements, undertakings, understandings, discussions, correspondence, or other documents or communications between the Party and any Released Person in relation to any of the foregoing (together, the "Claims Relating to Zim"); and

- (b) each Bond Trustee, on its own behalf and on behalf of the Bondholders in respect of which it is acting as trustee, hereby irrevocably, expressly and unconditionally:
 - (i) waives, releases, acquits and discharges to the fullest extent permissible by applicable law, any and all Claims (subject to any Claims arising by reason of fraud or deceit) that it ever had, now has or may have against any Released Person (in each case excluding any Bondholder-Selected Person);
 - (ii) agrees (subject to any Claims arising by reason of fraud or deceit), not to take or pursue any legal or other action in respect of any Claims against any Released Person (in each case excluding any Bondholder-Selected Person); and
 - (iii) agrees (subject to any Claims arising by reason of fraud or deceit), not to take or pursue any legal or other action in respect of any Claims against any Bondholder-Selected Person (other than:
 - (1) legal proceedings (but no other Enforcement Action) carried out in accordance with clauses 7.2 through 7.5 (inclusive) below; and

(2) Enforcement Action taken against a Bondholder-Selected Person or any of its assets to enforce a final, non-appealable judgment rendered against that Bondholder-Selected Person pursuant to legal proceedings prosecuted in compliance with clauses 7.2 through 7.5 (inclusive) below),

(in each case) upon, in connection with, or by reason of any Claims Relating to Zim.

- 7.2 (a) The waivers, releases, acquittals, discharges and any undertaking or agreement provided under clause 7.1 by (i) Zim or any Obligor to any Shipowner or Related Party; and (ii) by any Shipowner or Related Party to Zim or any Obligor (in each case) do not apply to:
 - (i) the items set out in clause 6.7(b) after the words "For the avoidance of doubt"; and
 - (ii) Claims to the extent arising under existing charter party arrangements (other than in respect of Deferred Hire), such as on-going ordinary course P&I claims of an operational nature involving Zim and vessels owned or operated by the Shipowners or Related Parties which (in each case) do not involve the payment of hire or accrued or deferred hire or callable exchange notes, in respect of the period prior to the Restructuring Effective Time and of which:
 - (A) (in respect of claims against the Group) the Shipowner or Related Party (as relevant) counterparty to any such existing charter party was unaware at the Restructuring Effective Time;
 - (B) (in respect of claims against Related Parties or Shipowners) the Group was unaware at the Restructuring Effective Time; and
 - (C) (in respect of Tynwald Navigation Limited and Marown Navigation Limited) Claims which are the subject of the settlement agreement referred to in Schedule 5 (*New Shipowner Documents*) (it being acknowledged that such Claims are to be settled in accordance with that agreement).
 - (b) Each person who is an Original Bondholder, and the Bond Trustees (on their own behalf and on behalf of the Original Bondholders in respect of which they are acting as trustee), have the right to commence legal proceedings in respect of the Claims Relating to Zim against Bondholder-Selected Persons only, but only subject to the undertakings, agreements and limited conditions set out below in clause 7.3. This right of the Bond Trustees and the Original Bondholders is personal to the Bond Trustees (on their own behalf and on behalf of the Original Bondholders in respect of which they are acting as trustee) and the Original Bondholders (respectively) and may not be charged, pledged, assigned, made the subject of a trust or similar arrangement or transferred (in each case in whole or in part), and any purported such action is void.
- 7.3 It is hereby agreed that (without derogating from any previous releases by and among the Bond Trustees, Bondholders and the Related Parties and their Connected Persons if and to the extent provided as part of the Company's 2009 restructuring), no legal proceedings may be commenced against a Bondholder-Selected Person by any Bond Trustee or Original Bondholder for any Claims Relating to Zim, unless the commencement of such legal proceedings is approved by

Original Bondholders holding at the Restructuring Effective Time at least 75% by face value of all the bonds (including series A, series B and series C) issued by the Company and outstanding immediately prior to the Restructuring Effective Time (the "Vote"); and subject to 5 Business Days' notice delivered by the bond trustee to the relevant Bondholder-Selected Persons (the "Notice"). For avoidance of all doubt, the above-mentioned mechanism is the sole and exclusive manner whatsoever to file any Claims Relating to Zim or for taking any actions against a Bondholder Selected Person under any circumstances in respect of Claims Relating to Zim. The Notice will include a list of each of the persons voting at the Original Bondholder assembly (the "Voting Persons") together with the results of such Vote. If at the relevant time of the Vote there shall exist a practical mechanism allowing a person, with no unreasonable efforts or costs, to obtain written evidence from a member of the TASE demonstrating that such person is an Original Bondholder, then any Original Bondholder participating in the Vote will be required to present such evidence prior to voting and as a condition to voting, and such evidence will be presented to the relevant Bondholder-Selected Person at its reasonable request.

- 7.4 In the event such proceedings have been initiated against a Bondholder-Selected Person:
 - (a) the "losing party" shall reimburse the "winning party" for all of its actual costs in connection with the proceedings; and
 - (b) (notwithstanding the releases in clauses 7.1) to the extent a Bondholder-Selected Person is entitled (by subrogation or otherwise) to pursue claims against the Company in respect of any such proceedings, such Bondholder-Selected Person shall not be precluded from pursuing such rights against the Company.
- 7.5 No assignees or transferees of the New Bondholder Documents and/or Series 1 Notes issued at the Restructuring Effective Time to Original Bondholders (but only to the extent such assignees or transferees are not themselves Original Bondholders), shall have any of the rights set out above and no such transferees or assignees (or any related Bond Trustee) shall initiate any legal proceedings against any of the Bondholder-Selected Persons.
- 7.6 For the purpose of identifying the Original Bondholders, the Bond Trustees (on behalf of the Bondholders) agree that the Company will deliver to the Related Parties a copy of any list of Bondholders delivered by the applicable Bond Trustee to the Company for the purposes of the Restructuring on or immediately prior to the Restructuring Effective Time and six months thereafter. Any Bondholder listed under any of those lists will be considered an Original Bondholder, notwithstanding the right of any person not included in those lists to otherwise provide evidence of its being an Original Bondholder. Such lists shall be delivered to the Related Parties within two Business Days from receiving it by the Company, from each Bond Trustee.
- 7.7 For the purposes of this clause 7:

"Bondholder-Selected Persons" means both (a) the Related Parties and their direct and indirect shareholders, and the direct and indirect Subsidiaries of the Related Parties and any of their respective employees, officers, directors and advisors; and (b) the direct and indirect shareholders of Millenium (but not Millenium itself and not any of Millenium's employees, officers, directors or advisors). Provided, however, that with respect to employees, officers, directors and advisors of the Related Parties, Claims Relating to Zim shall not include negotiations regarding the Restructuring set forth hereunder.

"Enforcement Action" means:

- (a) the making of any demand against, or the commencement of any legal proceedings against, any Bondholder-Selected Person (in each case) in relation to any Claim; or
- (b) the exercise of any right of set-off or counterclaim against any Bondholder-Selected Person (in each case) in respect of any Claim.

"Original Bondholder" means a Bondholder immediately prior to the Restructuring Effective Time as identified in Clause 7.6.

8. EFFECTIVENESS OF MUTUAL RELEASES

- 8.1 Clause 7 (Mutual Releases) shall not:
 - (a) affect the enforceability of the New Documents or any other transactions or agreements required to implement the Restructuring; or
 - (b) apply to any breaches of the New Documents arising in respect of events or circumstances arising after the Restructuring Effective Time; or
 - (c) apply to any Claims under this Deed.
- 8.2 No Party shall have any Claim against another Party (or any of that Party's Connected Persons) solely as a result of any Party's entry into this Deed or any of the documents related to the Restructuring or solely as a result of any Party's termination of its obligations under this Deed before the Restructuring Effective Time pursuant to clause 18.2 of this Deed.

9. TRANSFERS

No Party to this Deed may assign, novate, declare a trust over, or transfer (in whatever way or form) (any of which, a "transfer") any of its right, title, interest or economic risk in and to any arrangement, agreement or understanding (including any documents) relating to the Group prior to the Restructuring Effective Time unless simultaneously with such transfer the transferee (the "Transferee") accedes to this Deed in the same capacity as the transferor in accordance with clause 10 (Accession) and accedes to or otherwise executes any releases or New Documents in connection with the transactions the subject of this Deed that a Participating Stakeholder is otherwise required to execute as a condition precedent to the occurrence of the Restructuring Effective Time. Nothing in this clause 9 (Transfers) prohibits Bondholders from trading bonds to the extent not prohibited from trading by the TASE.

10. ACCESSION

- 10.1 Any Transferee shall at any time after the date of this Deed but before the Restructuring Effective Time become a Party to this Deed on the date that it delivers a duly executed and completed Accession Deed to the Company.
- 10.2 By delivering an Accession Deed in accordance with clause 10.1 above, such person shall be bound by, assume and comply with all the obligations under, and be entitled to all the rights under, all the terms of this Deed as if it had been a Party in the relevant capacity on and from the date of this Deed.

11. FURTHER ASSURANCES

After the Restructuring Effective Time, Zim or any Obligor shall, at its cost, promptly at the request of a Participating Stakeholder, execute and deliver such documents and do such other things, as may reasonably be required by any such Participating Stakeholder to give full effect to this Deed. Each Participating Stakeholder shall, promptly at the request of Zim and with that Party's reasonable costs incurred to be for the account of Zim, execute and deliver such documents and do such things, as may reasonably be required to give full effect to this Deed.

12. THIRD PARTY RIGHTS

- Save as otherwise expressly provided in this Deed, no person other than a party hereto shall have any right by virtue of the Third Parties Act to enforce any term (express or implied) of this Deed.
- 12.2 Notwithstanding clause 12.1, the Released Persons (including the Bondholder-Selected Persons as defined in clause 7 (*Mutual Releases*)) have the right under the Third Parties Act to enforce their rights under clause 7 (*Mutual Releases*) and each Party's Connected Persons has the right under the Third Parties Act to enforce its rights under clause 7 (*Mutual Releases*).
- 12.3 The terms of this Deed may be amended by the Parties without the consent of any person who is not a Party.

13. WAIVER

No course of dealing or the failure of any Party to enforce any of the provisions of this Deed shall in any way operate as a waiver of such provisions and shall not affect the right of such Party thereafter to enforce each and every provision of this Deed in accordance with its terms.

14. REMEDIES, WAIVERS AND AMENDMENTS

- 14.1 No course of dealing or failure to exercise, nor any delay in exercising, on the part of any Party or Participating Stakeholder, any right or remedy under any Existing Documents or document in relation to any Existing Document shall operate to impair such right or remedy or be construed as a waiver thereof and shall not affect the right of any Party or Participating Stakeholder to enforce each and every provision of the Existing Documents in accordance with its terms, nor shall any single or partial exercise of any right or remedy prevent any further or other exercise of such right or remedy or the exercise of any other right or remedy. The rights and remedies provided in this Deed are cumulative and not exclusive of any rights or remedies provided by law.
- 14.2 Except as otherwise provided for in any other provision in this Deed, no term of this Deed may be amended or waived without the consent of each Participating Stakeholder.

15. ENTIRE AGREEMENT

This Deed, together with the New Documents to which any Party is a party, constitutes the entire agreement between, and understanding of, the Parties with respect to the subject matter of this Deed and supersedes any prior written or oral agreements or arrangements between the Parties in relation thereto.

16. COUNTERPARTS

This Deed may be executed in any number of counterparts and all such counterparts taken together shall be deemed to constitute one and the same instrument. Delivery of a counterpart of this Deed by e-mail attachment or telecopy shall be an effective mode of delivery.

17. PARTIAL INVALIDITY

If, at any time, any provision of this Deed is or becomes illegal, invalid or unenforceable in any respect under any law of any jurisdiction, neither the legality, validity or enforceability of the remaining provisions nor the legality, validity or enforceability of such provision under the law of any other jurisdiction will in any way be affected or impaired.

18. RESERVATION OF RIGHTS AND TERMINATION

- 18.1 Unless expressly provided to the contrary, this Deed does not amend, vary or waive any Party's rights under the Existing Documents, or any Party's rights as creditors of any member of the Group unless and until the Restructuring Effective Time occurs (and then only to the extent provided under the terms of this Deed and the New Documents). Notwithstanding any other provision of this Deed, each Bondholder Trustee (for and on behalf of the Bondholders in respect of which it is acting as trustee), Secured Lender, VesselCo Party and Lender will continue to be entitled to receive interest on its outstanding indebtedness at the contractual rate (but not default interest or premia) up to (but excluding) the day on which the Restructuring Effective Time occurs. To the extent any such accrued interest is expressed to be payable in cash (but not otherwise) then the Company undertakes (to those parties entitled to receive that interest) to pay any such accrued and unpaid interest within 3 Business Days of the Restructuring Effective Time (or, in the case of cash interest due to the Bondholders, if later, at the earliest time permitted by the TASE for that interest to be paid).
- At any time before the Restructuring Effective Time (but not afterwards), any Party may terminate its obligations under this Deed by giving notice of termination (i) if by letter, when actually delivered to Zim; or (ii) by email to each of: danieli.rafi@il.zim.com; and eldar.guy@il.zim.com; with a copy to each of warnerp@sullcrom.com; davidm@friedman.co.il; and adva@gkh-law.com. Notwithstanding clause 20 (*Notices*), no other form of termination notice is effective unless such other form of termination notice has been actually confirmed by Zim to have been received. If this Deed is terminated by any Party for any reason, the rights of that Party against the other Parties to this Deed and those other Parties' rights against the terminating Party shall be fully reserved.
- 18.3 The Company will promptly inform the Parties if it receives a notice of termination under clause 18.2 from a Party.
- 18.4 The Company shall not issue the Restructuring Completion Letter following receipt of a notice of termination under clause 18.2 (unless that notice has been withdrawn).

19. PARTIES' RIGHTS AND OBLIGATIONS

19.1 The obligations of each Party under this Deed are several. Failure by a Party to perform its obligations under this Deed does not affect the obligations of any other Party under this Deed. No Party is responsible for the obligations of any other Party under this Deed.

19.2 The rights of each Party under or in connection with this Deed are separate and independent rights. A Party may separately enforce its rights under this Deed.

20. NOTICES

20.1 Communications in writing

Any communication to be made under or in connection with this Deed shall be made in writing and, unless otherwise stated, may be made by fax or letter.

20.2 Addresses

The address and fax number (and the department or officer, if any, for whose attention the communication is to be made) of each Party for any communication or document to be made or delivered under or in connection with this Deed is that identified in the signature pages to this Deed or any Deed of Accession or any substitute address, fax number or department or officer as the Party may notify to the other Parties by not less than five Business Days' notice, and provided that any change may be made to any notice details of any Party if such change is agreed by the Company and that Party.

20.3 Delivery

Any communication or document made or delivered by one person to another under or in connection with this Deed will only be effective:

- (a) if by way of fax, when received in legible form; or
- (b) if by way of letter, when it has been received at the relevant address,

and, if a particular department or officer is specified as part of its address details provided under clause 20.2 (*Addresses*), if addressed to that department or officer.

20.4 Electronic communication

- (a) Any communication to be made between the Parties under or in connection with this Deed may be made by electronic mail or other electronic means if the Parties:
 - (i) agree that, unless and until notified to the contrary, this is to be an accepted form of communication;
 - (ii) notify each other in writing of their electronic mail address and/or any other information required to enable the sending and receipt of information by that means; and
 - (iii) notify each other of any change to their address or any other such information supplied by them.

1

(b) Any electronic communication made between the Parties, will be effective only when actually received in readable form.

20.5 English language

- (a) Any notice given under or in connection with this Deed must be in English.
- (b) All other documents provided under or in connection with this Deed must be:
 - (i) in English; or
 - (ii) if not in English, and if so required by a Participating Stakeholder, accompanied by a certified English translation and, in this case, the English translation will prevail unless the document is a constitutional, statutory or other official document.

21. GOVERNING LAW

This Deed and any non-contractual obligations arising out of or in relation to this Deed are governed by English law, except for Schedule 15 (*Registration Rights*) of this Deed and any non-contractual obligations arising out of or in connection with that Schedule, which shall be interpreted in accordance with the laws of Israel (without prejudice to the fact that this Deed is governed by English law).

22. ENFORCEMENT

22.1 Jurisdiction

- (a) The courts of England have exclusive jurisdiction to settle any dispute arising out of or in connection with this Deed (including a dispute relating to the existence, validity or termination of this Deed or any non-contractual obligation arising out of or in connection with this Deed but excluding any dispute relating to Schedule 15 (*Registration Rights*), the jurisdiction of which shall be determined in accordance with that Schedule) (a "**Dispute**").
- (b) The Parties agree that the courts of England are the most appropriate and convenient courts to settle Disputes and accordingly no Party will argue to the contrary.
- (c) For the avoidance of doubt, the jurisdiction for enforcement of each New Document shall be as set out in that New Document.

22.2 Service of process

- (a) Without prejudice to any other mode of service allowed under any relevant law, Zim and each Obligor irrevocably appoints Ramon Insurance Brokers, 3rd Floor 24 Creechurch Lane, London EC3A 5EH as its agent for service of process in relation to any proceedings before the English courts in connection with this Deed.
- (b) Zim and each Obligor agrees that failure by a process agent to notify it of the process will not invalidate the proceedings concerned.

IN WITNESS WHEREOF, this Deed has been executed as a deed and delivered by the Parties on the date and year first above written.

SCHEDULE 1

OBLIGORS AND PARTICIPATING STAKEHOLDERS

Part I: Obligors

- 1. Darsal Shipping Inc (as borrower pursuant to the RBS Konza Facility)
- 2. Firoka Maritime Company Limited (as borrower pursuant to the RBS Firoka Facility)
- 3. Flamingo Navigation (S350) Company Ltd (as borrower under the BNPP Ksure Facility)
- 4. Flamingo Navigation (S351) Company Ltd (as borrower under the BNPP Ksure Facility)
- 5. Flamingo Navigation (S352) Company Ltd (as borrower under the Wilmington 352 Facility)
- 6. Kateland Navigation SA (as borrower pursuant to the RBS Kateland Facility)
- 7. Konza Shipping Ltd (as borrower pursuant to the RBS Konza Facility)
- 8. Koron Maritime Inc. (as guarantor pursuant to the RBS Firoka Facility)
- 9. Lympic Maritime Ltd (as borrower pursuant to the RBS Kateland Facility)
- 10. Pelican Maritime (S346) Company Ltd (as borrower under the BNPP Ksure Facility)
- 11. Pelican Maritime (S347) Company Ltd (as borrower under the BNPP Kexim Facility)
- 12. Pelican Maritime (S348) Company Ltd (as borrower under the BNPP Kexim Facility)
- 13. Pelican Maritime (S393) Company Ltd (as borrower under the BNPP Kexim Facility)
- 14. Pelican Maritime (S394) Company Ltd (as borrower under the BNPP Kexim Facility)
- 15. Pelican Maritime (S395) Company Ltd (as borrower under the BNPP Kexim Facility)
- 16. Pelican Maritime (S345) Company Ltd (as borrower under the Wilmington 345 Facility)
- 17. Pelican Maritime (S349) Company Ltd (as borrower under the Wilmington 349 Facility)
- 18. Sea Box Finance Marine Star (as borrower pursuant to the BNPP Sea Box Facility)
- 19.18. Yellow Sea Shipping Inc. (as guarantor pursuant to the KfW Bilateral Facility)
- 20.19. Ymir International Limited (as the owner of the vessel securing Zim Integrated Shipping Services Ltd.'s debt under the Series A, B and C bonds)
- 21.20. Zim Asia Maritime Company S.a. r.l. (as borrower under the former HSH Asia Facility)
- 22.21. Zim Asia Maritime Limited (as borrower under the HSH Asia Facility)
- 23.22. Zim Atlantic Maritime Company S.a. r.l. (as borrower under the former HSH Atlantic Facility)

- 24.23. Zim Atlantic Maritime Limited (as borrower under the HSH Atlantic Facility)
- 25.24. Zim Barcelona Maritime Company S.a. r.l. (as borrower under the <u>former HSH Barcelona</u> Facility)
- 26.25. Zim Barcelona Maritime Limited (as borrower under the HSH Barcelona Facility)
- 27.26. Zim China Maritime Company S.a. r.l. (as borrower under the <u>former HSH China Facility</u>)
- 28.27. Zim China Maritime Limited (as borrower under the HSH China Facility)
- 29.28. Zim Europa Maritime Company S.a. r.l. (as borrower under the <u>former HSH</u> Europa Facility)
- 30.29. Zim Europa Maritime Limited (as borrower under the HSH Europa Facility)
- 31.30. Zim Haifa Maritime Company S.a. r.l. (as borrower under the former HSH Asia Facility)
- 32.31. Zim Haifa Maritime Limited (as borrower under the HSH Asia Facility)
- 33.32. Zim Iberia Maritime Company S.a. r.l. (as borrower under the <u>former HSH</u> Iberia Facility)
- 34.33. Zim Iberia Maritime Limited (as borrower under the HSH Iberia Facility)
- 35.34. Zim Jamaica Maritime Company S.a. r.l. (as borrower under the <u>former HSH Jamaica Facility</u>)
- 36.35. Zim Jamaica Maritime Limited (as borrower under the HSH Jamaica Facility)
- 37.36. Zim Pacific Maritime Company S.a. r.l. (as borrower under the former HSH Pacific Facility)
- 38.37. Zim Pacific Maritime Limited (as borrower under the HSH Pacific Facility)

Part II: Secured Vessel Lenders

- 1. HSH Nordbank AG
- 2. Silver Chalk 4 Ltd.
- 3. The Royal Bank of Scotland plc

Part III: VesselCo Parties

BNPP Kexim Facility

- 1. ABN-AMRO Bank N.V. (as Syndicate Lender under the BNPP Kexim Facility)
- 2. Banco Popolare SC (as a Bank of America, National Association (as Syndicate Lender under the BNPP Kexim Facility)
- 3. Bank of America, National Association (as a Syndicate Lender under the BNPP Kexim Facility)
- 4.3. Belfius Bank S.A./N.V. (as a Syndicate Lender under the BNPP Kexim Facility)

- 5.4. BNP Paribas S.A. (as Facility Agent under the BNPP Kexim Facility)
- 6.5. BNP Paribas S.A. (as Syndicate Lender under the BNPP Kexim Facility)
- 7.6. Credit Industriel Et Commercial (as Syndicate Lender under the BNPP Kexim Facility)
- <u>8.7.</u> Natixis (as Syndicate Lender under the BNPP Kexim Facility)
- 9.8. The Export Import Bank Ofof Korea (as Syndicate Lender under the BNPP Kexim Facility)

BNPP Ksure Facility

- 1. ABN-AMRO Bank N.V. (as Syndicate Lender under the BNPP Ksure Facility)
- 2. Bank of Scotland plc (as Syndicate Lender under the BNPP Ksure Facility)
- 3. Belfius Bank S.A<u>./N.V</u>. (as Syndicate Lender under the BNPP Ksure Facility)
- 4. BNP Paribas S.A. (as Facility Agent under the BNPP Ksure Facility)
- 5. BNP Paribas S.A. (as Syndicate Lender under the BNPP Ksure Facility)
- 6. Credit Industriel Et Commercial (as Syndicate Lender under the BNPP Ksure Facility)
- 7. Natixis (as Syndicate Lender under the BNPP Ksure Facility)

Wilmington 345 Facility, Wilmington 349 Facility and Wilmington 352 Facility ("Wilmington Facilities")

- 1. Bank of America Merrill Lynch International Limited (as Syndicate Lender under the Wilmington Facilities)
- 2. Burlington Loan Management Limited (as Syndicate Lender under the Wilmington Facilities)
- 3. Citigroup Financial Products Inc. (as Syndicate Lender under the Wilmington Facilities)
- 4. King Street Acquisition Company, LLC (as Syndicate Lender under the Wilmington Facilities)
- 5. Wilmington Trust (London) Limited (as Facility Agent under the Wilmington Facilities)
- 6. York Global Finance BDH, LLC (as Syndicate Lender under the Wilmington Facilities)

Part IV: Shipowners

Conti Group

- 1. CONTI 151. Container Schiffahrts-GmbH & Co. KG MS "CONTI EMDEN"
- 2. CONTI 159. Container Schifffahrts-GmbH & Co. KG MS "CONTI SAN FRANCISCO"
- 3. CONTI 160. Container Schifffahrts-GmbH & Co. KG MS "CONTI ONTARIO"

Costamare Group

- 4. Alexia Transport Corp.
- 5. Angistri Corporation
- 6. Fastsailing Maritime Co.

Danaos Group

- 7. Balticsea Marine Inc.
- 8. Bayview Shipping Inc.
- 9. Blacksea Marine Inc.
- 10. Channelview Marine Inc.
- 11. Continent Marine Inc.
- 12. Medsea Marine Inc.

E.R. Schiffart Group

- 13. Twentysecond Dragon Shipping Inc.
- 14. Twentythird Dragon Shipping Inc.

F.A. Vinnen Group

- 15. Schifffahrtsgesellschaft "Merkur Archipelago" mbH & Co. KG
- 16. Schifffahrtsgesellschaft "Merkur Horizon" mbH & Co. KG

Funada

17. Fortune Line Inc.

NSC Group

- 18. Allocean Containers Limited
- 19. Allocean Containers No. 2 Limited

Rickmers Group

- 20. Marown Navigation Limited
- 21. Tynwald Navigation Limited

TDS Containerships

22. TDS Containerships IV AS

Zodiac Group

- 23. Berryford Shipping (UK) Limited
- 24. Bi-Level Shipping (UK) Limited
- 25. Carrion Maritime (UK) Limited
- 26. Jixi Maritime Limited

Part V: Bond Trustees

- 1. Hermetic Trust (1975) Ltd., bond trustee under Zim's Series C bonds (dated November 30th 2009 (as amended on July 11th, 2012))
- 2. Union Bank Trust Company Ltd., bond trustee under Zim's Series A bonds (dated November 30th 2009 as amended on July 11th 2012)
- 3. Union Bank Trust Company Ltd., bond trustee under Zim's Series B bonds (dated November 30th 2009 as amended on July 11th 2012)

Part VI: Lenders

- 1. Banca Nazionale del Lavoro SpA (BNL) Genova
- 2. BNP Paribas S.A. (Dublin Branch)
- 3. Citibank N.A
- 4. Citigroup Financial Products Inc.
- 5. Crown Managed Accounts SPC on behalf of and for the account of Crown/GLG Segregated Portfolio
- 6. European Distressed MAC Limited
- 7. GLG European Distressed Fund
- 8. GLG European Distressed Master Fund Ltd.
- 9. GLG European Distressed Master Fund II
- 10. GLG Market Neutral Fund
- 11. Israel Discount Bank Ltd.
- 12. Mercantile Discount Bank Ltd.
- 13. Union Bank of Israel Ltd.

14. Van Der Giessen- De Noord N.V.

Part VII: Related Parties

- 1. Danestead Shipping (UK) Limited
- 2. Sorla Shipping Inc.
- 3. Zipton Shipping (UK) Limited
- 4. Atwood Shipping Inc.
- 5. Derone Maritime Limited
- 6. Halton Maritime S.A.
- 7. Hurricane Maritime B.V. (The Netherlands)
- 8.7. Jakoby Maritime S.A.
- 9.8. Jazton Shipping Inc.
- 10.9. Liss Maritime Inc.
- 11.10. Pavan Shipping Ltd.
- 12.11. Stav Shipping Ltd.
- 13.12. Tacton Shipping Inc.
- 14.13. XT Maritime Ltd.

Part VIII: HHI Parties

- 1. Hyundai Heavy Industries Co., Ltd
- 2. Hyundai Samho Heavy Industries Co., Ltd.

I

SCHEDULE 2

OUTSTANDINGS AND ALLOCATION TABLE

Stakeholder	Outstanding principal debt as at 30/6/2014 ¹	Summary of restructuring Arrangements				Total claims / allocation as at Restructuring Effective Time		
	(\$ '000)	2	Total Deficiency claim immediately prior to Restructuring Effective Time (\$ '000)	Allocation of Series 1, Series 2 and Equity pursuant to Deficiency Claim as at Restructuring Effective Time (\$ '000)			Total Debt claims as at Restructuring Effective Time (Tranche A + Series 1 + Series 2) (\$ '000)	9
		,		Series 1	Series 2	Equity		
Secured Vessel Lenders								
HSH Nordbank AG ³	335,279	103,500	231,779	79,278	-	11.746%	182,778	11.746%
Zim Asia	31,342	7,500	23,842	7,830	-	1.160%	15,330	1.160%
Zim Atlantic	31,342	7,500	23,842	7,830	-	1.160%	15,330	1.160%
Zim Haifa	55,458	23,500	31,958	11,884	-	1.761%	35,384	1.761%
Zim Barcelona	54,464	23,500	30,964	11,589	-	1.717%	35,089	1.717%
Zim Pacific	31,342	7,500	23,842	7,830	-	1.160%	15,330	1.160%
Zim Jamaica	32,833	8,500	24,333	8,079	-	1.197%	16,579	1.197%
Zim Europa	32,833	8,500	24,333	8,079	-	1.197%	16,579	1.197%
Zim Iberia	32,833	8,500	24,333	8,079	-	1.197%	16,579	1.197%
Zim China	32,833	8,500	24,333	8,079	-	1.197%	16,579	1.197%
Silver Chalk 4 Ltd (Known as King Street Bilateral) ⁴	30,103	8,500	21,603	7,271	-	1.077%	15,771	1.077%
Zim USA	30,103	8,500	21,603	7,271	-	1.077%	15,771	1.077%

This does not include any applicable legal, financial advisory and out-of-pocket expenses previously agreed to be reimbursed by the Group.

² For VesselCo Parties this is the amount allocated to the VesselCo arrangements.

References in rows under the heading "HSH-Nordbank AG" are to facilities financing the vessels named in those rows.

⁴ References in the row under the heading "Silver Chalk 4 Ltd (Known as King Street Bilateral)" are to the facilities financing the vessels named in that row.

Stakeholder	Outstanding principal debt as at 30/6/2014 ¹	Summary of rest	ructuring Arrang	gements	summary of restructuring Arrangements				
	(\$ '000)		Total Deficiency claim immediately prior to Restructuring Effective Time (\$ '000)				Total Debt claims as at Restructuring Effective Time (Tranche A + Series 1 + Series 2) (\$ '000)	Total Equity as at Restructuring Effective Time	
		,		Series 1	Series 2	Equity			
The Royal Bank of Scotland Plc ⁵	152,412	104,675	47,737	24,933	-	3.694%	129,608	3.694%	
Zim Yokohama ⁶	36,806	25,000	11,806	6,075	-	0.900%	31,075	0.900%	
Novorossiysk Star	9,272	6,375	2,897	1,515	-	0.225%	7,890	0.225%	
Zim Vancouver ⁷	36,321	25,000	11,321	5,931	-	0.879%	30,931	0.879%	
Zim Qingdao ⁸	34,271	23,300	10,971	5,652	-	0.837%	28,952	0.837%	
Zim Shekou ⁹	35,742	25,000	10,742	5,760	-	0.853%	30,760	0.853%	
VesselCo Lenders									
BNP Paribas S.A. Kexim Facility	200,968	-	23,622	10,324	-	1.530%	10,324	1.530%	
The Export Import Bank Of Korea	148,295	-	17,431	7,618	-	1.129%	7,618	1.129%	
BNP Paribas S.A.	7,282	-	856	374	-	0.055%	374	0.055%	
Bank of America N.A., National Association	965 12,812	-	113 1,506	50 658	_	0. 007 <u>098</u> %	50 658	0.007098%	
ABN-AMRO Bank N.V.	11,847	-	1,393	609	-	0.090%	609	0.090%	
Credit Industriel et Commercial	8,885	-	1,044	456	-	0.068%	456	0.068%	

⁵ References in rows under the heading "The Royal Bank of Scotland plc" are to facilities financing the vessels named in those rows.

The amount set out in this row for Zim Yokohama together with the amount set out in the row entitled "Zim Vancouver" represents in aggregate the amount for the facility agreement between The Royal Bank of Scotland plc, Konza Shipping Ltd. and Darsal Shipping Ltd. dated 28 August 2006 as amended or supplemented from time to time.

The amount set out in this row for Zim Vancouver together with the amount set out in the row entitled "Zim Yokohama" represents in aggregate the amount for the facility agreement between The Royal Bank of Scotland plc, Konza Shipping Ltd. and Darsal Shipping Ltd. dated 28 August 2006 as amended or supplemented from time to time.

The amount set out in this row for Zim Qingdao together with the amount set out in the row entitled "Zim Shekou" represents in aggregate the amount for the facility agreement between The Royal Bank of Scotland plc, Kateland Navigation S.A. and Lympic Maritime Limited dated 4 May 2006 as amended or supplemented from time to time.

The amount set out in this row for Zim Shekou together with the amount set out in the row entitled "Zim Qingdao" represents in aggregate the amount for the facility agreement between The Royal Bank of Scotland plc, Kateland Navigation S.A. and Lympic Maritime Limited dated 4 May 2006 as amended or supplemented from time to time.

Stakeholder	Outstanding principal debt as at 30/6/2014 ¹	Summary of rest					Total claims / a Restructuring	Effective Time
	(\$ '000)	Reinstated debt on Tranche A terms as at Restructuring Effective Time ²	Total Deficiency claim immediately prior to Restructuring Effective Time (\$ '000)		ries 1, Series 2 an aim as at Restruc	d Equity pursuant turing Effective	Total Debt claims as at Restructuring Effective Time (Tranche A + Series 1 + Series 2) (\$ '000)	Total Equity as at Restructuring Effective Time
				Series 1	Series 2	Equity		
Natixis	5,924	-	696	304	-	0.045%	304	0.045%
Belfius Bank S.A <u>./N.V</u> .	5,924	-	696	304	-	0.045%	304	0.045%
Banco Popolare SC	11,847	-	1,393	609	-	0.090%	609	0.090%
BNP Paribas S.A. Ksure Facility	254,368	-	29,618	14,067	-	2.084%	14,067	2.084%
BNP Paribas S.A.	76,694	-	8,930	4,241	-	0.628%	4,241	0.628%
Bank of Scotland plc	47,052	-	5,479	2,602	-	0.386%	2,602	0.386%
ABN-AMRO Bank N.V.	47,499	-	5,531	2,627	-	0.389%	2,627	0.389%
Credit Industriel et Commercial	35,624	-	4,148	1,970	-	0.292%	1,970	0.292%
Natixis	23,750	-	2,765	1,313	-	0.195%	1,313	0.195%
Belfius Bank S.A <u>./N.V</u> .	23,750	-	2,765	1,313	-	0.195%	1,313	0.195%
Wilmington Facility ¹⁰	274,007	-	49,257	26,041	_	3.858%	26,041	3.858%
Citigroup Financial Products, Inc.	33,964	-	7,419	3,922	-	0.581%	3,922	0.581%
King Street Acquisition Company, LLC	129,911	-	23,293	12,314	-	1.824%	12,314	1.824%
Burlington Loan Management Limited	89,399		15,054	7,959		1.179%	7,959	1.179%
Bank of America Merrill Lynch International Limited	5,000		842	445		0.066%	445	0.066%
York Global Finance BDH, LLC	15,733	-	2,650	1,401	-	0.208%	1,401	0.208%
Wilmington 345 Facility	98,274		17,666	9,340		1.384%	9,340	1.384%
Citigroup Financial Products, Inc.	12,081	-	2,661	1,407	-	0.208%	1,407	0.208%
King Street Acquisition Company, LLC	46,598	-	8,354	4,417	-	0.654%	4,417	0.654%
Burlington Loan Management Limited	32,141	-	5,399	2,854	-	0.423%	2,854	0.423%
Bank of America Merrill Lynch International Limited	1,798	-	302	160	-	0.024%	160	0.024%
York Global Finance BDH, LLC	5,656	-	950	502	-	0.074%	502	0.074%

This row sets out the aggregate amount for each of the Wilmington 345, 349 and 352 Facilities.

Stakeholder	Outstanding principal debt as at 30/6/2014 ¹	Summary of rest	ructuring Arrang	gements			Total claims / a Restructuring	
	(\$ '000)	Reinstated debt on Tranche A terms as at Restructuring Effective Time ²	Total Deficiency claim immediately prior to Restructuring Effective Time (\$ '000)	Allocation of Series 1, Series 2 and Equity pursuant to Deficiency Claim as at Restructuring Effective Time (\$ '000)			Total Debt claims as at Restructuring Effective Time (Tranche A + Series 1 + Series 2) (\$ '000)	
	Ź				Series 2	Equity		
Wilmington 352 Facility	86,052		15,469	8,178		1.212%	8,178	1.212%
Citigroup Financial Products, Inc.	10,883	-	2,330	1,232	-	0.182%	1,232	0.182%
King Street Acquisition Company, LLC	40,788	-	7,315	3,867	-	0.573%	3,867	0.573%
Burlington Loan Management Limited	27,908	-	4,728	2,499	-	0.370%	2,499	0.370%
Bank of America Merrill Lynch International Limited	1,561	-	264	140	-	0.021%	140	0.021%
York Global Finance BDH, LLC	4,911	-	832	440	-	0.065%	440	0.065%
Wilmington 349 Facility	89,682		16,122	8,523		1.263%	8,523	1.263%
Citigroup Financial Products, Inc.	10,999	-	2,428	1,284	-	0.190%	1,284	0.190%
King Street Acquisition Company, LLC	42,525	-	7,624	4,030	-	0.597%	4,030	0.597%
Burlington Loan Management Limited	29,351	-	4,927	2,605	-	0.386%	2,605	0.386%
Bank of America Merrill Lynch International Limited	1,642	-	276	146	-	0.022%	146	0.022%
York Global Finance BDH, LLC	5,165	-	867	458	-	0.068%	458	0.068%
Shipowners ¹¹								
Conti Reederei Management GmbH & Konzeptions KG ¹²	65,260	-	65,260	3,380	15,776	2.838%	19,156	2.838%
Conti 151.	1,503	-	1,503	78	363	0.065%	441	0.065%
Conti 159.	31,386	-	31,386	1,626	7,587	1.365%	9,213	1.365%
Conti 160.	32,371	-	32,371	1,677	7,825	1.408%	9,502	1.408%
Costamare Inc.	28,036	-	28,036	1,452	6,777	1.219%	8,229	1.219%

This was calculated according to the agreed formulation in Jerusalem, which consists of the sum of (1) the present value of future charter hire deficiency discounted at a rate of 8%; and (2) the outstanding principal amount of CENs plus accrued PIK interest.

The names listed below are abbreviations of the full legal names of the entities which are listed in the endnotes to this Schedule.

Stakeholder	Outstanding principal debt as at 30/6/2014 ¹	Summary of restr	ructuring Arrang				Total claims / a Restructuring	Effective Time
	(\$ '000)	Reinstated debt on Tranche A terms as at Restructuring Effective Time ² (\$ '000)	Total Deficiency claim immediately prior to Restructuring Effective Time (\$ '000)	Allocation of Ser to Deficiency Cla Time (\$ '000)		d Equity pursuant turing Effective	Total Debt claims as at Restructuring Effective Time (Tranche A + Series 1 + Series 2) (\$ '000)	
				Series 1	Series 2	Equity		
Alexia Transport Corp.	9,247	-	9,247	479	2,235	0.402%	2,714	0.402%
Angistri Corporation	9,293	-	9,293	481	2,246	0.404%	2,728	0.404%
Fastsailing Maritime Co.	9,495	-	9,495	492	2,295	0.413%	2,787	0.413%
Danaos Corporation	169,996	-	169,996	8,804	41,095	7.393%	49,899	7.393%
Balticsea Marine Inc.	28,097	-	28,097	1,455	6,792	1.222% 1.190%	8,247	1.222%
Bayview Shipping Inc. Blacksea Marine Inc.	27,369	-	27,369	1,417	6,616	1.190%	8,034	1.190% 1.276%
Channelview Marine Inc.	29,343	-	29,343	1,520	7,093		8,613	
Continent Marine Inc.	27,877 28,427	-	27,877 28,427	1,444 1,472	6,739 6,872	1.212% 1.236%	8,183	1.212% 1.236%
Medsea Marine Inc.	28,883	-	28,883	1,472	6,982	1.256%	8,344 8,478	1.256%
E.R. Schiffahrt GmbH & Co. KG	20,184	-	20,184	1,045	4,879	0.878%	5,924	0.878%
Twentysecond Dragon Shipping Inc.	9,571	-	9,571	496	2,314	0.416%	2,809	0.416%
Twentythird Dragon Shipping Inc.	10,613	_	10,613	550	2,566	0.462%	3,115	0.462%
F.A. Vinnen & Co (GmbH & Co KG) ¹³	40,150	-	40,150	2,079	9,706	1.746%	11,785	1.746%
Merkur Archipalego	20,062	-	20,062	1,039	4,850	0.872%	5,889	0.872%
Merkur Horizon	20,088	-	20,088	1,040	4,856	0.874%	5,896	0.874%
Funada	44,365	-	18,002	932	4,352	0.783%	5,284	0.783%
Fortune Line Inc. 14	44,365	-	18,002	932	4,352	0.783%	5,284	0.783%
NSC Allocean Containers Limited	42,304	-	42,304	2,191	10,227	1.840%	12,418	1.840%
Allocean Containers Limited	21,127	-	21,127	1,094	5,107	0.919%	6,201	0.919%
Allocean Containers No. 2 Limited	21,178	-	21,178	1,097	5,119	0.921%	6,216	0.921%

The names listed below are abbreviations of the full legal names of the entities which are listed in the endnotes to this Schedule.

The amount under the column headed "Total Deficiency claim" is less than the amount under the heading "Outstanding principal debt" due to the fact that the vessel the subject of this claim was structured as a finance lease prior to the Restructuring Effective Time.

Stakeholder	Outstanding principal debt as at 30/6/2014 ¹	Summary of rest						Total claims / allocation as at Restructuring Effective Time		
		Reinstated debt on Tranche A terms as at Restructuring Effective Time ² (\$ '000)	Total Deficiency claim immediately prior to Restructuring Effective Time (\$ '000)	Allocation of Ser to Deficiency Cla Time (\$ '000)		d Equity pursuant turing Effective	Total Debt claims as at Restructuring Effective Time (Tranche A + Series 1 + Series 2) (\$ '000)	Total Equity as at Restructuring Effective Time		
				Series 1	Series 2	Equity				
Rickmers Holding GmbH & Cie. KG ¹⁵	46,740	-	46,740	2,421	11,299	2.033%	13,720	2.033%		
Marown Navigation Limited	23,279	-	23,279	1,206	5,627	1.012%	6,833	1.012%		
Tynwald Navigation Limited	23,461	-	23,461	1,215	5,671	1.020%	6,886	1.020%		
Thor Dahl	2,576	-	2,576	133	623	0.112%	756	0.112%		
TDS Containerships IV A.S.	2,576	-	2,576	133	623	0.112%	756	0.112%		
Zodiac Maritime Agencies Ltd.	53,492	-	40,741	2,110	9,849	1.772%	11,959	1.772%		
Bi-Level Shipping (UK) Limited	3,480	-	3,480	180	841	0.151%	1,022	0.151%		
Berryford Shipping (UK) Limited	2,745	-	2,745	142	663	0.119%	806	0.119%		
Jixi Maritime Limited	9,925	-	9,925	514	2,399	0.432%	2,913	0.432%		
Carrion Maritime (UK) Limited ¹⁶	37,343	-	24,591	1,274	5,945	1.069%	7,218	1.069%		
Bondholders										
Union Bank Trust Company Ltd. (Series A)	179,431	20,856	158,575	58,724	-	8.701%	79,580	8.701%		
Union Bank Trust Company Ltd. (Series B)	72,919	8,476	64,444	23,865	-	3.536%	32,341	3.536%		
Hermetic Trust (1975) Ltd. (Series C)	169,218	19,669	149,549	55,382	-	8.205%	75,050	8.205%		
Lenders										
Union Bank of Israel Ltd.	1,295	177	1,118	414	-	0.061%	591	0.061%		
Citigroup Financial Products Inc.	13,306	1,838	11,468	4,247	-	0.629%	6,085	0.629%		

This excludes amounts payable to the Rickmers Holding GmbH & Cie. KG group under settlement and related arrangements under the documents referred to under the heading "Rickmers Group" in Schedule 5 (copies of which have been posted in the VDR).

The amount under the column headed "Total Deficiency claim" is less than the amount under the heading "Outstanding principal debt" due to the fact that the vessel the subject of this claim was structured as a finance lease prior to the Restructuring Effective Time.

Stakeholder	Outstanding principal debt as	Summary of rest	ructuring Arrang	gements			Total claims / allocation as at Restructuring Effective Time		
	at 30/6/2014 ¹						Restructuring	Effective Time	
		Reinstated debt	Total	Allocation of Ser	ries 1, Series 2 an	d Equity pursuant	Total Debt	Total Equity	
		on Tranche A	Deficiency	to Deficiency Cla	aim as at Restruc	cturing Effective	claims as at	as at	
		terms as at	claim	Time			Restructuring	Restructuring	
		Restructuring	immediately				Effective Time	Effective	
		Effective Time ²	prior to				(Tranche A +	Time	
			Restructuring				Series 1 +		
			Effective Time				Series 2)		
	(\$ '000)	(\$ '000)	(\$ '000)	(\$ '000)	1	_	(\$ '000)		
					Series 2	Equity			
GLG Partners	4,555	625	3,930	1,456	-	0.216%	2,080	0.216%	
GLG European Distressed Fund	1,458	200	1,257	466		0.069%	0.07%	0.069%	
GLG European Distressed Master Fund Ltd.	1,184	163	1,022	379		0.056%	0.06%	0.056%	
European Distressed MAC Limited	729	100	629	233		0.035%	0.03%	0.035%	
GLG Market Neutral Fund	592	81	511	189		0.028%	0.03%	0.028%	
Crown Managed Accounts SPC on behalf of	592	81	511	189		0.028%	0.03%	0.028%	
and for the account of Crown/GLG Segregated Portfolio									
Banca Nazionale del Lavoro SpA (BNL)	673	93	580	215	_	0.032%	308	0.032%	
Genova	073	73	300	213		0.03270	300	0.03270	
BNP Paribas S.A. (Dublin branch)	7,353	1,010	6,343	2,349	-	0.348%	3,359	0.348%	
Citibank N.A	6,403	882	5,521	2,045	-	0.303%	2,927	0.303%	
Israel Discount Bank Ltd.	1,237	172	1,065	394	-	0.058%	566	0.058%	
Mercantile Discount Bank Ltd.	13,692	1,876	11,816	4,376	-	0.648%	6,252	0.648%	
Van der Giessen - De Noord N.V.	13,558	1,854	11,705	4,335	-	0.642%	6,188	0.642%	
GLG European Distressed Master Fund II	413	-	413	121	-	0.018%	121	0.018%	
Related Parties									
Israel Corporation Ltd.	225,150 228,218	-	225,150 <u>228,218</u>	-	-		-	[32.00 %] % 17	
Millenium Investments Elad Ltd.	50,550	-	50,550	-	-		-	-	
Eastern Pacific Shipping (UK) Limited Ltd.	65,735	-	51,081	-	-	-	-	-	
Danestead Shipping (UK) Limited	3,605	-	3,605	-	-	-	-	-	
Sorla Shipping Inc.	16,395	-	16,395	-	-	-	-	-	
Zipton Shipping (UK) Ltd ¹⁸	45,735	-	31,081	-	-	_	-	_	

This allocation of equity to IC arises pursuant to the cash subscription by IC to Zim of \$200 million.

Stakeholder	Outstanding principal debt as at 30/6/2014 ¹	-	·					Total claims / allocation as at Restructuring Effective Time		
		Reinstated debt on Tranche A terms as at Restructuring Effective Time ²	Total Deficiency claim immediately prior to Restructuring Effective Time (\$ '000)	to Deficiency Claim as at Restructuring Effective Time			Total Debt claims as at Restructuring Effective Time (Tranche A + Series 1 + Series 2) (\$ '000)			
	(4 333)	(4 000)	(4 333)	Series 1	Series 2	Equity	(+)			
XT Holdings	165,222 162,154	-	134,910 131,841	-	-		-	-		
XT Maritime Ltd	6,200	-	6,200	-	-	-	-	-		
Liss Maritime Inc.	2,597	-	2,597	-	-	-	-	-		
Halton Maritime S.A.	3,810	-	3,810	-	-	-	-	-		
Jakoby Maritime S.A.	13,184	-	13,184	-	-	-	-	-		
Derone Maritime Ltd	14,408	-	14,408	-	-	-	-	-		
Tacton Shipping Inc. ¹⁹	60,934	-	45,944	-	-	-	-	-		
Jazton Shipping Inc. ²⁰	61,021	-	45,699	-	-	-	-	-		
Hurricane Maritime B.V. (The Netherlands)	3,068	-	3,068	-	-	-	-	-		
Other Vessel Debt										
Hyundai Samho Heavy Industries Co., Ltd ²¹	92,654	-	92,654	27,197	-		27,197			
Hyundai Heavy Industries Co., Ltd										

The amount under the column headed "Total Deficiency claim" is less than the amount under the heading "Outstanding principal debt" due to the fact that the vessel the subject of this claim was structured as a finance lease prior to the Restructuring Effective Time.

The amount under the column headed "Total Deficiency claim" is less than the amount under the heading "Outstanding principal debt" due to the fact that the vessel the subject of this claim was structured as a finance lease prior to the Restructuring Effective Time.

The amount under the column headed "Total Deficiency claim" is less than the amount under the heading "Outstanding principal debt" due to the fact that the vessel the subject of this claim was structured as a finance lease prior to the Restructuring Effective Time.

Hyundai Samho Heavy Industries Co., Ltd will receive the HHI Subordinated Loan Agreement in place of equity in Zim as part of the restructuring in the amount of \$65,457,000.

i CONTI 151. Container Schiffahrts-GmbH & Co. KG MS "CONTI EMDEN"

ii CONTI 159. Container Schifffahrts-GmbH & Co. KG MS "CONTI SAN FRANCISCO"

iii CONTI 160. Container Schifffahrts-GmbH & Co. KG MS "CONTI ONTARIO"

iv Schifffahrtsgesellschaft "Merkur Archipelago" mbH & Co. KG

v Schifffahrtsgesellschaft "Merkur Horizon" mbH & Co. KG

NEW SVL DOCUMENTS¹

HSH Nordbank AG

Gene	General				
No.	Document				
1.	Global Restructuring Deed				
2.	Special Account Release				
3.	Waiver Fee Deferral Letter				

Zim	Zim Asia						
No.	Document						
1.	Loan Agreement in respect of "Zim Asia"						
<u>2.</u>	Amendment and Restatement Agreement in respect of the Loan Agreement in respect of "Zim						
	Asia"						
<u>3.</u>	Shortfall Loan Agreement in respect of "Zim Asia" ²						
2 4.	Amendment Agreement in respect of the Shortfall Loan Agreement in respect of "Zim _Asia"						
3 5.	Security Documents relevant to "Zim Asia"						

Zim	Zim Atlantic					
No.	Document					
1.	Loan Agreement in respect of "Zim Atlantic"					
<u>2.</u>	Amendment and Restatement Agreement in respect of the Loan Agreement in respect of "Zim					
	Atlantic"					
<u>3.</u>	Shortfall Loan Agreement in respect of "Zim Atlantic"					
2 4.	Amendment Agreement in respect of the Shortfall Loan Agreement in respect of "Zim					
	Atlantic" ⁵					
<u>35</u> .	Security Documents relevant to "Zim Atlantic"					

Zim	Zim Barcelona						
No.	Document						
1.	Loan Agreement in respect of "Zim Barcelona"						
<u>2.</u>	Amendment and Restatement Agreement in respect of the Loan Agreement in respect of "Zim						

This does not include amortisation deferral/ covenant waiver fees of approximately US\$ 3.4 million in aggregate owing to the Secured Vessel Lenders and the VesselCo Parties and payable in July 2015. These amounts are reflected in the June 2014 Business Plan.

² To be discharged at the Restructuring Effective Time.

³ To be discharged at the Restructuring Effective Time.

⁴ To be discharged at the Restructuring Effective Time.

⁵ To be discharged at the Restructuring Effective Time.

	Barcelona "
<u>3.</u>	Shortfall Loan Agreement in respect of "Zim Barcelona"
2 4.	Amendment Agreement in respect of the Shortfall Loan Agreement in respect of "Zim
	Barcelona",7
3 <u>5</u> .	Security Documents relevant to "Zim Barcelona"

Zim	Zim China	
No.	Document	
1.	Loan Agreement in respect of "Zim China"	
<u>2.</u>	Amendment and Restatement Agreement in respect of the Loan Agreement in respect of "Zim	
	China"	
<u>3.</u>	Shortfall Loan Agreement in respect of "Zim China" ⁸	
2 4.	Amendment Agreement in respect of the Shortfall Loan Agreement in respect of "Zim China"	
3 <u>5</u> .	Security Documents relevant to "Zim China"	

Zim	Zim Europa	
No.	Document	
1.	Loan Agreement in respect of "Zim Europa"	
<u>2.</u>	Amendment and Restatement Agreement in respect of the Loan Agreement in respect of "Zim	
	Europa"	
<u>3.</u>	Shortfall Loan Agreement in respect of "Zim Europa" 10	
2 4.	Amendment Agreement in respect of the Shortfall Loan Agreement in respect of "Zim	
	Europa'' ¹¹	
<u>35</u> .	Security Documents relevant to "Zim Europa"	

Zim	Zim Haifa	
No.	Document	
1.	Loan Agreement in respect of "Zim Haifa"	
<u>2.</u>	Amendment and Restatement Agreement in respect of the Loan Agreement in respect of "Zim	
	Haifa"	
<u>3.</u>	Shortfall Loan Agreement in respect of "Zim Haifa" 12	
2 4.	Amendment Agreement in respect of the Shortfall Loan Agreement in respect of "Zim Haifa" Haifa	
<u>35</u> .	Security Documents relevant to "Zim Haifa"	

To be discharged at the Restructuring Effective Time.

To be discharged at the Restructuring Effective Time.

⁸ To be discharged at the Restructuring Effective Time.

 $^{{\}color{red} \frac{12}{}}$ To be discharged at the Restructuring Effective Time.

Zim	Zim Iberia	
No.	Document	
1.	Loan Agreement in respect of "Zim Iberia"	
<u>2.</u>	Amendment and Restatement Agreement in respect of the Loan Agreement in respect of "Zim	
	<u>Iberia"</u>	
<u>3.</u>	Shortfall Loan Agreement in respect of "Zim Iberia" 14	
2 4.	Amendment Agreement in respect of the Shortfall Loan Agreement in respect of "Zim	
	Iberia" ¹⁵	
<u>35</u> .	Security Documents relevant to "Zim Iberia"	

Zim	Zim Jamaica	
No.	Document	
1.	Loan Agreement in respect of "Zim Jamaica"	
<u>2.</u>	Amendment and Restatement Agreement in respect of the Loan Agreement in respect of "Zim	
	Jamaica"	
<u>3.</u>	Shortfall Loan Agreement in respect of "Zim Jamaica" 16	
2 4.	Amendment Agreement in respect of the Shortfall Loan Agreement in respect of "Zim	
	Jamiaca" Jamaica"	
3 <u>5</u> .	Security Documents relevant to "Zim Jamaica"	

Zim	Zim Pacific	
No.	Document	
1.	Loan Agreement in respect of "Zim Pacific"	
<u>2.</u>	Amendment and Restatement Agreement in respect of the Loan Agreement in respect of "Zim Pacific"	
<u>3.</u>	Shortfall Loan Agreement in respect of "Zim Pacific" 18	
2 4.	Amendment Agreement in respect of the Shortfall Loan Agreement in respect of "Zim Pacific" Pacific".	
3 <u>5</u> .	Security Documents relevant to "Zim Pacific"	

Silver Chalk 4 Ltd.

To be discharged at the Restructuring Effective Time.

To be discharged at the Restructuring Effective Time.

 $[\]underline{^{16}} \quad \text{To be discharged at the Restructuring Effective Time.}$

¹⁷— To be discharged at the Restructuring Effective Time.

 $[\]underline{^{18}}$ — To be discharged at the Restructuring Effective Time.

¹⁹—To be discharged at the Restructuring Effective Time.

Gen	General	
No.	Document	
1.	Global Restructuring Deed	
2.	Special Account Release	
3.	Waiver Fee Deferral Letter	

Zim	Zim USA	
No.	Document	
1.	Amendment and Restatement Agreement in respect of "Zim USA"	
2.	Amended guarantee	
3.	Security Documents relevant to "Zim USA"	

The Royal Bank of Scotland plc

Gen	General	
No.	Document	
1.	Global Restructuring Deed	
2.	Special Account Release	
3.	Waiver Fee Deferral Letter	

Novo	Novorossiysk Star	
No.	Document	
1.	Amendment and Restatement Agreement in respect of "Novorossiysk Star"	
2.	Amended Zim parent guarantee	
3.	Security Documents relevant to "Novorossiysk Star"	

Zim	Zim Qingdao and Zim Yokohama	
No.	Document	
1.	Amendment and Restatement Agreement in respect of "Zim Qingdao" and "Zim Yokohama"	
2.	Amended Zim parent guarantee	
3.	Security Documents relevant to "Zim Qingdao" and "Zim Yokohama"	

Zim	Zim Vancouver and Zim Shekou	
No.	Document	
1.	Amendment and Restatement Agreement in respect of "Zim Vancouver" and "Zim Shekou"	
2.	Amended Zim parent guarantee	
3.	Security Documents relevant to "Zim Vancouver" and "Zim Shekou"	
4.	Stock transfer forms	
5.	Share Pledge	

NEW VESSELCO PARTY DOCUMENTS¹

BNPP Kexim Facility (comprising ABN-AMRO Bank N.V. (as Syndicate Lender); Banco Popolare SC (as Syndicate Lender); Bank of America, National Association (as Syndicate Lender); Belfius Bank S.A./N.V. (as Syndicate Lender); BNP Paribas S.A. (as Facility Agent); BNP Paribas S.A. (as Syndicate Lender); Credit Industriel et Commercial (as Syndicate Lender); Natixis (as Syndicate Lender); The Export-Import Bank of Korea (as Syndicate Lender))

Gen	General BNPP Kexim Syndicate Documents	
No.	Document	
1.	Global Restructuring Deed	
2.	Overall Agreement	
3.	Release Documents	
4.	Waiver Fee Deferral Letter	
<u>5.</u>	Container Sale and Purchase Agreement (and related Bill of Sale)	
<u>6.</u>	Container Lease Agreement	
<u>7.</u>	Release of existing container pledge	
<u>8.</u>	Israeli release and termination letter to insurers regarding the assignment of insurances	
	under the existing Israeli container pledge	

Zim	Zim Rotterdam	
No.	Document	
1.	Share Sale and Purchase Agreement	
2.	Time Charter Agreement	
<u>3.</u>	Tripartite Agreement (and related Management Agreement)	
<u>4.</u>	Cancellation of existing bareboat charter agreements	
<u>5.</u>	Assignment of any sub-charters entered into by Zim from time to time (and equivalent Hebrew	
	<u>Debenture</u>)	
<u>6.</u>	Sub-Charter Earnings Account Charge (and equivalent Hebrew Debenture)	
<u>7.</u>	Israeli Vessel Debenture (and equivalent Hebrew Debenture)	
<u>38</u> .	Acknowledgement Acknowledgments of notice notices of assignment assignments of Time	
	Charter Agreement Agreements	

Zim	Zim Tianjin	
No.	Document	
1.	Share Sale and Purchase Agreement	
2.	Time Charter Agreement	
<u>3.</u>	Tripartite Agreement (and related Management Agreement)	
4.	Cancellation of existing bareboat charter agreements	

This does not include amortisation deferral/ covenant waiver fees of approximately US\$ 3.4 million in aggregate owing to the Secured Vessel Lenders and the VesselCo Parties and payable in July 2015. These amounts are ——reflected in the June 2014 Business Plan.

<u>5.</u>	Assignment of any sub-charters entered into by Zim from time to time (and equivalent Hebrew
	<u>Debenture</u>)
<u>6.</u>	Sub-Charter Earnings Account Charge (and equivalent Hebrew Debenture)
<u>7.</u>	Israeli Vessel Debenture (and equivalent Hebrew Debenture)
<u>38</u> .	Acknowledgement Acknowledgments of notice notices of assignment assignments of Time
	Charter Agreement Agreements

BNPP Ksure Facility (comprising ABN-AMRO Bank N.V. (as Syndicate Lender); Bank of Scotland plc (as Syndicate Lender); Belfius Bank S.A./N.V. (as Syndicate Lender); BNP Paribas S.A. (as Facility Agent); BNP Paribas S.A. (as Syndicate Lender); Credit Industriel et Commercial (as Syndicate Lender); Natixis (as Syndicate Lender))

Gene	General BNPP Ksure Syndicate Documents	
No.	Document	
1.	Global Restructuring Deed	
2.	Overall Agreement	
3.	Release Documents	
4.	Waiver Fee Deferral Letter	

Zim	Zim Antwerp	
No.	Document	
1.	Share Sale and Purchase Agreement	
2.	Time Charter Agreement	
3.	Deed of release from HHI	
4.	Acknowledgement of notice of assignment of Time Charter Agreement	
5.	Termination of Intercreditor Deed	

Zim	Zim Ningbo	
No.	Document	
1.	Share Sale and Purchase Agreement	
2.	Time Charter Agreement	
3.	Deed of release from HHI	
4.	Acknowledgement of notice of assignment of Time Charter Agreement	
5.	Termination of Intercreditor Deed	

Zim	Zim San Diego	
No.	Document	
1.	Share Sale and Purchase Agreement	
2.	Time Charter Agreement	
3.	Deed of release from HHI	
4.	Acknowledgement of notice of assignment of Time Charter Agreement	
5.	Termination of Intercreditor Deed	

<u>Wilmington 345 Facility</u> (comprising Bank of America Merrill Lynch International Limited (as Syndicate Lender); Burlington Loan Management Limited (as Syndicate Lender); Citigroup Financial Products Inc. (as Syndicate Lender); King Street Acquisition

Company L.L.C. (as Syndicate Lender); Wilmington Trust (London) Limited (as Facility Agent); York Global Finance BDH, LLC (as Syndicate Lender))

Zim	Zim Djibouti	
No.	Document	
1.	Global Restructuring Deed	
2.	Release Documents	
3.	Special Account Release	
4.	Memorandum of Agreement	
5.	Time charterparty agreement	
6.	Deed of release from HHI	
7.	Waiver Fee Deferral Letter	
8.	Termination of Intercreditor Deed	

<u>Wilmington 349 Facility</u> (comprising Bank of America Merrill Lynch International Limited (as Syndicate Lender); Burlington Loan Management Limited (as Syndicate Lender); Citigroup Financial Products Inc. (as Syndicate Lender); King Street Acquisition Company L.L.C. (as Syndicate Lender); Wilmington Trust (London) Limited (as Facility Agent); York Global Finance BDH, LLC (as Syndicate Lender))

Zim	Zim Los Angeles	
No.	Document	
1.	Global Restructuring Deed	
2.	Release Documents	
3.	Special Account Release	
4.	Memorandum of Agreement	
5.	Time charterparty agreement	
6.	Deed of release from HHI	
7.	Waiver Fee Deferral Letter	
8.	Termination of Intercreditor Deed	

<u>Wilmington 352 Facility</u> (comprising Bank of America Merrill Lynch International Limited (as Syndicate Lender); Burlington Loan Management Limited (as Syndicate Lender); Citigroup Financial Products Inc. (as Syndicate Lender); King Street Acquisition Company L.L.C. (as Syndicate Lender); Wilmington Trust (London) Limited (as Facility Agent); York Global Finance BDH, LLC (as Syndicate Lender))

Zim	Zim Chicago	
No.	Document	
1.	Global Restructuring Deed	
2.	Release Documents	
3.	Special Account Release	
4.	Memorandum of Agreement	
5.	Time charterparty agreement	
6.	Waiver Fee Deferral Letter	

NEW SHIPOWNER DOCUMENTS

Conti Group

CON	CONTI 151. Container Schiffahrts-GmbH & Co. KG MS "CONTI EMDEN"	
No.	Document	
1.	Global Restructuring Deed	

	CONTI 159. Container Schifffahrts-GmbH & Co. KG MS "CONTI SAN FRANCISCO"	
No.	Document	
1.	Global Restructuring Deed	
2.	Existing charter	
3.	Charter Addendum	
4.	Calculation Agency Agreement	

CON	CONTI 160. Container Schifffahrts-GmbH & Co. KG MS "CONTI ONTARIO"	
No.	Document	
1.	Global Restructuring Deed	
2.	Existing charter	
3.	Charter Addendum	
4.	Calculation Agency Agreement	

Costamare Group

Alex	Alexia Transport Corp.	
No.	Document	
1.	Global Restructuring Deed	
2.	Existing charter	
3.	Charter Addendum	
4.	Calculation Agency Agreement	

Angi	Angistri Corporation	
No.	Document	
1.	Global Restructuring Deed	
2.	Existing charter	
3.	Charter Addendum	
4.	Calculation Agency Agreement	

Fast	Fastsailing Maritime Co.	
No.	Document	
1.	Global Restructuring Deed	
2.	Existing charter	
3.	Charter Addendum	

Fast	Fastsailing Maritime Co.	
No.	Document	
4.	Calculation Agency Agreement	

Danaos Group

Balti	Balticsea Marine Inc.	
No.	Document	
1.	Global Restructuring Deed	
2.	Existing charter	
3.	Charter Addendum	
4.	Calculation Agency Agreement	

Bayv	Bayview Shipping Inc.	
No.	Document	
1.	Global Restructuring Deed	
2.	Existing charter	
3.	Charter Addendum	
4.	Calculation Agency Agreement	

Blac	Blacksea Marine Inc.	
No.	Document	
1.	Global Restructuring Deed	
2.	Existing charter	
3.	Charter Addendum	
4.	Calculation Agency Agreement	

Cha	Channelview Marine Inc.	
No.	Document	
1.	Global Restructuring Deed	
2.	Existing charter	
3.	Charter Addendum	
4.	Calculation Agency Agreement	

Cont	Continent Marine Inc.	
No.	Document	
1.	Global Restructuring Deed	
2.	Existing charter	
3.	Charter Addendum	
4.	Calculation Agency Agreement	

Med	Medsea Marine Inc.	
No.	Document	
1.	Global Restructuring Deed	
2.	Existing charter	

Med	Medsea Marine Inc.	
No.	No. Document	
3.	Charter Addendum	
4.	Calculation Agency Agreement	

E.R. Schiffahrt Group

Twe	Twentysecond Dragon Shipping Inc.	
No.	Document	
1.	Global Restructuring Deed	
2.	Existing charter	
3.	Charter Addendum	
4.	Calculation Agency Agreement	

Twe	Twentythird Dragon Shipping Inc.	
No.	Document	
1.	Global Restructuring Deed	
2.	Existing charter	
3.	Charter Addendum	
4.	Calculation Agency Agreement	

F.A. Vinnen & Co Group

Schi	Schifffahrtsgesellschaft "Merkur Archipelago" mbH & Co. KG	
No.	Document	
1.	Global Restructuring Deed	
2.	Existing charter	
3.	Charter Addendum	
4.	Calculation Agency Agreement	

Schi	Schifffahrtsgesellschaft "Merkur Horizon" mbH & Co. KG	
No.	Document	
1.	Global Restructuring Deed	
2.	Existing charter	
3.	Charter Addendum	
4.	Calculation Agency Agreement	

Fortune Line Inc.

No.	Document
1.	Global Restructuring Deed
2.	Existing charter
3.	Charter Addendum

NSC Allocean Containers Group

Allo	Allocean Containers Limited	
No.	Document	
1.	Global Restructuring Deed	
2.	Existing charter	
3.	Charter Addendum	
4.	Calculation Agency Agreement	

Allo	Allocean Containers No. 2 Limited	
No.	Document	
1.	Global Restructuring Deed	
2.	Existing charter	
3.	Charter Addendum	
4.	Calculation Agency Agreement	

Rickmers Group

Mar	Marown Navigation Limited	
No.	Document	
1.	Global Restructuring Deed	
2.	Existing charter	
3.	Charter Addendum	
4.	Settlement Agreement	
5.	Retrofit side letter	

Tynv	Tynwald Navigation Limited	
No.	Document	
1.	Global Restructuring Deed	
2.	Existing charter	
3.	Charter Addendum	
4.	Settlement Agreement	
5.	Retrofit side letter	

TDS Containerships IV AS

No.	Document
1.	Global Restructuring Deed

Zodiac Group

Berr	Berryford Shipping (UK) Limited	
No.	Document	
1.	Global Restructuring Deed	

Bi-L	Bi-Level Shipping (UK) Limited	
No.	Document	
1.	Global Restructuring Deed	

Carı	Carrion Maritime (UK) Limited	
No.	Document	
1.	Global Restructuring Deed	
2.	Existing charter	
3.	Charter Addendum	

Jixi 1	Jixi Maritime Limited	
No.	Document	
1.	Global Restructuring Deed	
2.	Existing charter	
3.	Charter Addendum	

NEW BONDHOLDER DOCUMENTS

Hermetic Trust (1975) Ltd. (as bond trustee under Zim's Series C bonds)

No.	Document
1.	Global Restructuring Deed
2.	Restructuring Plan
3.	Series D Trust Deed
4.	Security Documents

Union Bank Trust Company Ltd (as bond trustee under Zim's Series A bonds)

No.	Document
1.	Global Restructuring Deed
2.	Restructuring Plan
3.	Series D Trust Deed
4.	Security Documents

Union Bank Trust Company Ltd (as bond trustee under Zim's Series B bonds)

No.	Document
1.	Global Restructuring Deed
2.	Restructuring Plan
3.	Series D Trust Deed
4.	Security Documents

NEW LENDER DOCUMENTS

Banc	Banca Nazionale del Lavoro SpA (BNL) Genova	
No.	Document	
1.	Global Restructuring Deed	
2.	Amendment and Restatement Agreement	
3.	Security Documents	

BNP	BNP Paribas S.A. (Dublin Branch)	
No.	Document	
1.	Global Restructuring Deed	
2.	Amendment and Restatement Agreement	
3.	Security Documents	

Citib	Citibank N.A	
No.	Document	
1.	Global Restructuring Deed	
2.	Amendment and Restatement Agreement	
3.	Security Documents	

Citig	Citigroup Financial Products Inc.	
No.	Document	
1.	Global Restructuring Deed	
2.	Amendment and Restatement Agreement	
3.	Security Documents	

Crov	Crown Managed Accounts SPC on behalf of and for the account of Crown/GLG Segregated	
Porti	Portfolio, European Distressed MAC Limited, GLG European Distressed Fund, GLG European	
Distr	Distressed Master Fund Ltd., GLG Market Neutral Fund	
No.	Document	
1.	Global Restructuring Deed	
2.	Amendment and Restatement Agreement	
3.	Security Documents	

GLG	GLG European Distressed Master Fund II	
No.	Document	
1.	Global Restructuring Deed	
2.	Deed of Release	

Israe	Israel Discount Bank Ltd.	
No.	Document	
1.	Global Restructuring Deed	
2.	Amendment and Restatement Agreement	
3.	Security Documents	

Merc	Mercantile Discount Bank Ltd.	
No.	Document	
1.	Global Restructuring Deed	
2.	Amendment and Restatement Agreement	
3.	Security Documents	

Unio	Union Bank of Israel Ltd.	
No.	Document	
1.	Global Restructuring Deed	
2.	Amendment and Restatement Agreement	
3.	Security Documents	

Van	Van Der Giessen- De Noord N.V.	
No.	Document	
1.	Global Restructuring Deed	
2.	Amendment and Restatement Agreement	
3.	Security Documents	

NEW IC DOCUMENTS

Israe	Israel Corporation Ltd	
No.	Document	
1.	Investment Agreement	
2.	Global Restructuring Deed	
3.	Contingent Claims Agreement	
4.	IC Framework Credit Line Agreement (Receivables Facility)	

NEW MILLENIUM DOCUMENTS

Millenium

Mille	Millenium Investments Elad Ltd.	
No.	Document	
1.	Global Restructuring Deed	
2.	Deed of Release	

NEW RELATED PARTIES DOCUMENTS

EPS Group

Dan	Danestead Shipping (UK) Limited	
No.	Document	
1.	Global Restructuring Deed	

Sorl	Sorla Shipping Inc.	
No.	Document	
1.	Global Restructuring Deed	
2.	Existing charter	
3.	Charter Addendum	

Zipt	Zipton Shipping (UK) Limited	
No.	Document	
1.	Global Restructuring Deed	
2.	Existing charter	
3.	Charter Addendum	

XT Group*

Atw	Atwood Shipping Inc.	
No.	Document	
1.	Global Restructuring Deed	

Dero	Derone Maritime Limited	
No.	Document	
1.	Global Restructuring Deed	
2.	Existing charter	
3.	Charter Addendum	

Halt	Halton Maritime S.A.	
No.	Document	
1.	Global Restructuring Deed	

Hurricane Maritime B.V.		
No.	Document	
1.	Global Restructuring Deed	

Jakoby Maritime S.A.			
No.	Document		
1.	Global Restructuring Deed		
2.	Existing charter		
3.	Charter Addendum		

Jazto	Jazton Shipping Inc.			
No.	Document			
1.	Global Restructuring Deed			
2.	Existing charter			
3.	Charter Addendum			

Liss Maritime Inc.		
No.	Document	
1.	Global Restructuring Deed	

Pava	Pavan Shipping Ltd.		
No.	Document		
1.	Global Restructuring Deed		

Stav Shipping Ltd.		
No.	Document	
1.	Global Restructuring Deed	

Tacton Shipping Inc.				
No.	Document			
1.	Global Restructuring Deed			
2.	Existing charter			
3.	Charter Addendum			

XT Maritime Ltd.				
No.	Document			
1.	Global Restructuring Deed			
2.	Deed of release of loan			

^{*}In addition to the above arrangements, the XT Maritime group will continue to provide ship management and operational services for the Zim Vancouver and Zim Shekou on commercial terms consistent with past practice for a period of up to 90 days after the Restructuring Effective Time.

NEW HHI PARTIES DOCUMENTS

Hyundai Heavy Industries Co. Ltd		
No.	Document	
1.	Global Restructuring Deed	
2.	Termination of Intercreditor Deed in respect of Zim Antwerp (BNPP Ksure)	
3.	Termination of Intercreditor Deed in respect of Zim Ningbo (BNPP Ksure)	
4.	Termination of Intercreditor Deed in respect of Zim San Diego (BNPP Ksure)	
5.	Termination of Intercreditor Deed in respect of Zim Antwerp (Wilmington)	
6.	Termination of Intercreditor Deed in respect of Zim Los Angeles (Wilmington)	

Hyundai Samho Heavy Industries Co. Ltd			
No.	Document		
1.	HHI Loan Agreement		
2.	Global Restructuring Deed		
3.	Deed of Release for Zim Antwerp (BNPP Ksure)		
4.	Release of Mortgage for Zim Antwerp (BNPP Ksure)		
5.	Reassignment of insurance for Zim Antwerp (BNPP Ksure)		
	Reassignment of earnings for Zim Antwerp (BNPP Ksure)		
6.	Termination of Intercreditor Deed in respect of Zim Antwerp (BNPP Ksure)		
7.	Deed of Release for Zim Ningbo (BNPP Ksure)		
8.	Release of Mortgage for Zim Ningbo (BNPP Ksure)		
9.	Reassignment of insurance for Zim Ningbo (BNPP Ksure)		
	Reassignment of earnings for Zim Ningbo (BNPP Ksure)		
10.	Termination of Intercreditor Deed in respect of Zim Ningbo (BNPP Ksure)		
11.	Deed of Release for Zim San Diego (BNPP Ksure)		
12.	Release of Mortgage for Zim San Diego (BNPP Ksure)		
13.	Reassignment of insurance for Zim San Diego (BNPP Ksure)		
	Reassignment of earnings for Zim San Diego (BNPP Ksure)		
14.	Termination of Intercreditor Deed in respect of Zim San Diego (BNPP Ksure)		
15.	Deed of Release for Zim Djibouti (Wilmington)		
16.	Release of Mortgage for Zim Djibouti (Wilmington)		
17.	Reassignment of insurance for Zim Djibouti (Wilmington)		
	Reassignment of earnings for Zim Djibouti (Wilmington)		
18.	Termination of Intercreditor Deed in respect of Zim Antwerp (Wilmington)		
19.	Deed of Release for Zim Los Angeles (Wilmington)		
20.	Release of Mortgage for Zim Los Angeles (Wilmington)		
21.	Reassignment of insurance for Zim Los Angeles (Wilmington)		
	Reassignment of earnings for Zim Los Angeles (Wilmington)		
22.	Termination of Intercreditor Deed in respect of Zim Los Angeles (Wilmington)		

ACCESSION DEED

THIS DEED OF ACCESSION is made on	by [name and details of acceding
stakeholder /creditor] (the "Acceding Party") in favour of ea	ach of the parties to the Global Restructuring
Deed (defined below).	

WHEREAS:

- (A) This Deed is supplemental to a global restructuring deed (the "Global Restructuring Deed") dated [●] and entered into by, among others, Zim Integrated Shipping Services Limited (the "Company"), certain members of the Company's group as Obligors, the Bond Trustees, the Lenders, the Secured Vessel Lenders, the VesselCo Parties, IC, Millenium, Hyundai Samho Heavy Industries Co., Ltd and other Participating Stakeholders (each as defined therein).
- (B) The Acceding Party wishes to accede to the Global Restructuring Deed as a [Bond Trustee][Lender] [Related Party] [Secured Vessel Lender] [VesselCo Party][Shipowner][Participating Stakeholder].
- (C) It is a term of the Restructuring Deed that in order to accede to it in the capacity above, the Acceding Party must enter into this Deed and deliver a duly executed and completed of this Deed to the Company

IT IS HEREBY AGREED AS FOLLOWS:

- 1. Words and expressions used by not defined herein shall have the meaning given in the Global Restructuring Deed.
- 2. The Acceding Party agrees, as of the date of this Deed, to be bound by all the terms and conditions of the Global Restructuring Deed insofar as they relate to a [Bond Trustee][Lender] [Party] [Secured Vessel Lender] [VesselCo Party][Shipowner][Participating Stakeholder] as if the Acceding Party were an original Party to the Global Restructuring Deed in such capacity with rights, obligations and interests as a [Bond Trustee][Lender] [Related Party] [Secured Vessel Lender] [VesselCo Party][Shipowner][Participating Stakeholder].
- 3. Any notice or other communication required to be given to the Acceding Party under the Global Restructuring Deed shall be sent to it at the address or email specified below or such other address or email as the Acceding Party may subsequently notify to Company.
 - [Name][Address][email address]
- 4. This Deed and any non-contractual obligations arising out of or in relation to this Deed are governed by English law.
- 5. The courts of England have exclusive jurisdiction to settle any dispute arising out of or in connection with this Deed (including a dispute relating to the existence, validity or termination of this Deed or any non-contractual obligation arising out of or in connection with this Deed) (a "Dispute").

6. The Acceding Party agrees that the courts of England are the most appropriate and convenient courts to settle Disputes and accordingly it shall not argue to the contrary.

IN WITNESS WHEREOF, this Deed has been executed as a deed and delivered by the Acceding Party on the date and year first above written.

SUBSCRIPTION LETTER

FORM OF DECLARATION

To:	Zim In	Zim Integrated Shipping Services Ltd. (the "Company")		
From:	M		(41 - 117	
		:(the "Investor")	
		SS:		
	C	tration Number:		
Compa	any; or (the Company of any of (i) ordinary shares of the under a trust indenture between the Company and igned confirms as follows.	
connec descril	ction wi	ith a distribution of ordinary shares and	18, as amended (the "Securities Law"), and in Wor Series 1 and 2 Notes of the Company (as rdance with the implementation of a restructuring resigned, hereby declare as follows:	
		rporation organized under the laws of [delete as necessary] rk either (a) or (b) below:	/ I am an individual, resident of	
			er the First Supplement of the Securities Law, by	
()		_	definitions and I am aware of the implications of	
	the s	status of a Classified Investor specified in	the First Supplement of the Securities Law and I	
	conse	ent thereto (please check the relevant boxe	es):	
			on that was incorporated for the purpose of the ing, whose equity exceeds NIS 50 million;	
		A joint investment mutual fund, as def 1994, or a managing company for such a	ined in the Israeli Joint Investment Trust Law of a fund;	
		A provident fund or its managing comp Services Law (Provident Funds) of 2005	any, as defined in the Israeli Control of Financial	
		An insurer, as defined under the Israeli	Insurance Business (Control) Law of 1981;	
		(Licensing) Law of 1981 - with the exce	y corporation, as defined in the Israeli Banking eption of joint services companies - purchasing for its who fall within the categories listed in section	
		Advice, Investment Marketing and Inve	tion 8(b) of the Israeli Regulation of Investment estment Portfolio Management Law of 1995, who ients who are investors that are listed in section	

	An investment advisor or investment marketer as defined in section 7(c) of the Israeli Regulation of Investment Advice and Portfolio Management Law of 1995, purchasing for himself / herself;
	A stock exchange member purchasing for itself or for clients who are investors that are listed in section 15A(B) of the Securities Law;
	An underwriter with respect to whom the qualifications prescribed in section 56(c) of the Securities Law have been met, purchasing for itself;
	A venture capital fund; for this purpose, a "venture capital fund" shall mean a corporation whose main business is investing in corporations, which, at the time the investment is made, are primarily engaged in research and development or in the manufacture of innovative and high-tech products or processes, where the risk of investment is higher than what is customary for other investments;
	A corporation which is wholly owned by investors listed in section $15A(B)$ of the Securities Law ⁴² ;
	An individual who complies with at least two of the following three requirements detailed below, purchasing for himself / herself:
	[Please indicate YES or NO with respect to each requirement] (a) I own cash, deposits, Financial Assets ⁴³ and Securities ⁴⁴ in total worth of more than NIS 12 million: YES / NO
	(b) I have expertise and skills in the capital market. My expertise is based on the following:
	Or alternatively, I was employed for a period of at least one year at in the following role, which is a professional role that requires expertise in capital market. YES / NO
	(c) During the four calendar quarters prior to the date hereof, I executed at least 30 transactions on average, in each quarter during the last four quarters, not including transactions performed on my behalf by my portfolio manager(s): YES / NO
(b) I am	not qualified as a "Classified Investor" under the First Supplement of the Securities Law.
3. Please chec	ck one of the following boxes:
	vestors listed in section 15A(B) of the Securities Law include, among other things, investors ring with at least one of the requirements detailed in this section 2(a).
43 See 'Definit	tions' Appendix
44 See 'Definit	ions' Appendix

l

	I purchase the Securities according to my own understanding and for my own benefit and on my own account and not with the aim or intention of distributing or offering them to other parties
	☐ I hold the Securities for the benefit of
4.	I undertake that until the Company becomes a public company, any transfer of the Securities (in whole or in part), as far as it is permitted by the Restructuring documents, and the Company's articles of association, either directly or indirectly, will be made to investors who comply with either section $15A(b)(1)$ or $15A(b)(2)$ of the Securities Law as shall be amended from time to time, and provided such investor declared to such in writing and in any event I will not transfer the Securities (in whole or in part) to anyone to whom such transfer requires the Company to publish a prospectus.
5.	I am not, and am not acting on behalf of anyone that is, in the United States and I am subscribing for the Securities in an offshore transaction in accordance with Regulation S under the U.S. Securities Act of 1933.
6.	I acknowledge that I can bear the economic risk and complete loss of my investment in the Securities and have such knowledge and experience in financial or business matters that enables me to evaluate the merits and risks of the investment contemplated hereby.
7.	I am aware that the Company will rely on this declaration.
	I undertake to inform the Company of any change to the declarations made in this declaration. ours sincerely,
Na	me: Date:
Sig	gnature:
Dاے	ease send via e-mail a scanned signed conv of this "Subscription Letter" (including any anney if

Please send via e-mail a scanned signed copy of this "Subscription Letter" (including any annex, if relevant) to Yuval Eden – yuvale@gkh-law.com and Tal Gat – talg@gkh-law.com. Subsequently deliver two (2) copies of the original signed documents by courier to:

Yuval Eden, Adv.

Gross, Kleinhendler, Hodak, Halevy, Greenberg & Co.
One Azrieli Center, Round Building
Tel Aviv 6701101, Israel

DEFINITIONS:

"Securities": including securities which are not included in the definition in section 1 of the Securities Law, and including units of a closed fund within the meaning of the Joint Investment Trust Law. The definition of "Securities" in section 1 of the Securities Law is: certificates issues in series by a company, a cooperative society or any other corporation conferring a right of membership or participation in them or claim against them, and certificates conferring a right to acquire securities, all of which whether registered or bearer securities, excluding securities issued by the Israeli Government or the Bank of Israel which comply with one of the following:

- (1) They do not confer a right of participation or membership in a corporation and are not convertible into, or realizable for, securities conferring such a right;
- (2) They are issued under special legislation.

"Financial Assets": are defined in the Advice Law to include units in mutual investment funds, shares or units of funds registered outside of Israel, options, future contracts, structured products and education funds (kranot hishtalmut).

RESTRUCTURING COMPLETION LETTER

From:	Zim Integrated Shipping Services Limited ("Zim")
To:	The Parties (as defined in the Global Restructuring Deed (as defined below), other than Zim)
Date:] 2014

This is the "Restructuring Completion Letter" referred to in the global restructuring deed dated on or about the date hereof between Zim and the parties named therein (the "Global Restructuring Deed"). Capitalised terms not otherwise defined in this letter have the meanings given to them in the Global Restructuring Deed.

Global Restructuring Deed

1. All Parties have signed the Global Restructuring Deed.

Shipowner Documents

- 2. We refer to the New Shipowner Documents.
- 3. We confirm that the New Shipowner Documents have been executed by all the parties thereto and that all conditions precedent to the effectiveness of the New Shipowner Documents (other than the issuance of this letter) have been satisfied.

Tranche A and Bond documents

- 4. We refer to the New SVL Documents, the New Bondholder Documents and the New Lender Documents.
- 5. We confirm that we have received confirmation from or on behalf of the relevant counterparties to those documents that all conditions precedent to the effectiveness of the New SVL Documents, the New Bondholder Documents and the New Lender Documents (as applicable) (other than the issuance of this letter) have been satisfied.

VesselCo documents

- 6. We refer to the New VesselCo Party Documents.
- 7. We confirm that we have received confirmation from or on behalf of the relevant counterparties to those documents that all conditions precedent to the effectiveness of the New VesselCo Party Documents (other than the issuance of this letter) have been satisfied.
- 8. We confirm that we have been advised by counsel to the lenders under the Wilmington 345 Facility, the Wilmington 349 Facility and the Wilmington 352 Facility that all relevant Liberian formalities associated with completion of the transfers of vessels under those arrangements have been completed.

Other "New Documents"

- 9. We refer to all other New Documents not included in the above.
- 10. We confirm that all conditions precedent to the effectiveness of any such New Documents (other than the issuance of this letter) have been satisfied.

Series 1 Notes and Series 2 Notes

- 11. We refer to the trust indenture constituting the Series 1 Notes and Series 2 Notes.
- 2. We confirm that the Series 1 Notes and Series 2 Notes have been duly issued and delivered by Zim and duly authenticated by the trustee in accordance with the indenture in an aggregate principal amount of \$371,600,000 (in respect of the Series 1 Notes) and \$114,600,000 (in respect of the Series 2 Notes).) have (in the case of Participating Stakeholders entitled to Series 1 Notes or Series 2 Notes who have delivered Subscription Letters and Settlement Instructions) been duly issued in the name of, and delivered by the Company to, The Registration Company of Bank Le'umi Le'Israel Ltd., as depository for the global notes for the Nesher System or TACT System (as relevant), together with instructions to credit the securities accounts in a TASE member held by or on behalf of the Participating Stakeholders in accordance with their instructions, and in such amounts as are consistent with the respective allocations of such Participating Stakeholders in Schedule 2 to the Global Restructuring Deed; or (in the case of such Participating Stakeholders who have not delivered Subscription Letters and Settlement Instructions at the date hereof) have been allocated to such Participating Stakeholders and will be issued promptly after receipt of the Subscription Letter and Settlement Instructions.

Articles

- 13. As at the date hereof, the articles of the Company are in the form set out in agreed by the Company's virtual data room as at June 30, 2014 which Parties and are effective as at this time pursuant to an order of the Israeli District Court in Haifa.
- 14. As at the date hereof, no court has issued an interim or final order restraining the Company from entering into any of the arrangements contemplated in the Global Restructuring Deed.

Equity

15. Attached as Schedule 1 is an extract of the shareholders' register of Zim showing the shareholders who have been allocated fully paid up shares in Zim at the time of issuance of this letter. This allocation of equity amongst the shareholders of Zim matches the allocation of equity in Zim set out in the Outstandings and Allocation Table (on the assumption that all Participating Stakeholders subscribe for equity (or their Designated Recipients have delivered a Subscription Letter to Zim)) except where noted on Schedule 1 in respect of persons who have not at the time of this letter delivered a Subscription Letter to Zim.

Receivables facility

16. We confirm that the Company and Israel Corporation have executed the \$50 million receivables facility with Israel Corporation as lender in the form most recently posted in the VDR and that the conditions precedent to funding under that facility are as set out in the conditions precedent satisfaction letter for that facility posted in the VDR.

Equity injection

17. We confirm that we have received the sum of \$200 million in cash in freely available funds from Israel Corporation by way of equity contribution. Attached as Schedule 2 is confirmation from our bank showing the crediting of our bank account with \$200 million received from Israel Corporation.

Board of directors

- 18. We confirm that each of the directors of Zim as at the date hereof (prior to the issuance of this letter) has delivered irrevocable resignation letters to us and that upon the issuance of this letter our board of directors will be as set out in Schedule 16 to the Global Restructuring Deed.
- 19. The following insurance arrangements have been put in place:
 - run-off insurance for liability of the Group's directors and officers for the seven-year period beginning at the Restructuring Effective Time, with a coverage limit of \$50 million and a total premium for the entire period of up to \$1 million; and
 - (b) a framework agreement for liability insurance for the Group's directors and officers for the period beginning at the Restructuring Effective Time and ending on the later of the fifth anniversary of the Restructuring Effective Time and the day of the Company's annual general meeting in 2019 to approve the Company's annual financial statements for the year 2018. That framework agreement will comprise annual policies on a claimsmade basis with a coverage amount of \$50 million and annual premiums not exceeding \$120,000.

This letter and any non-contractual arrangements arising out of or in connection with it are governed by English law.

Yours truly

Zim Integrated Shipping Services Limited

SCHEDULE 1 (TO THE RESTRUCTURING COMPLETION LETTER)

Shareholders' register

SCHEDULE 2 (TO THE RESTRUCTURING COMPLETION LETTER)

Zim Confirmation

SCHEDULE 15

REGISTRATION RIGHTS

In connection with the issuance of equity to each of the persons listed as a shareholder of the Company pursuant to the subscription under the Global Restructuring Deed, to which this document is scheduled (which may be referred to in this Schedule as the "**Original Holders**" and each may be referred to as an "**Original Holder**") as part of the debt reorganization of the Company, the Company and the Original Holders hereby agree that, *inter alia*, this Schedule shall govern the rights of the Holders to cause the Company to register with the U.S. Securities and Exchange Commission (the "**SEC**") or an equivalent authority in the event of a listing outside the U.S., the Ordinary Shares that are issuable to, held by or may hereafter be issued to the Holders; and

NOW, THEREFORE, the parties hereby agree as follows:

1. Definitions. For purposes of this schedule:

"Affiliate" means, with respect to any specified Person, any other Person who, directly or indirectly, controls, is controlled by, or is under common control with such Person, including without limitation any general partner, managing member, officer or director of such Person or any venture capital fund now or hereafter existing that is controlled by one or more general partners or managing members of, or shares the same management company with, such Person.

"Damages" means any loss, damage, claim or liability (joint or several) to which a party hereto may become subject under the Securities Act, the Exchange Act, or other federal or state law, insofar as such loss, damage, claim or liability (or any action in respect thereof) arises out of or is based upon: (i) any untrue statement or alleged untrue statement of a material fact contained in any registration statement of the Company, including any preliminary prospectus or final prospectus contained therein, any reasonable related free writing prospectus (as defined in Rule 405 of the Securities Act) or any amendments or supplements thereto and any related road show; (ii) an omission or alleged omission to state therein a material fact required to be stated therein, or necessary to make the statements therein not misleading; or (iii) any violation or alleged violation by the indemnifying party (or any of its agents or Affiliates) of the Securities Act, the Exchange Act, any state securities law, or any rule or regulation promulgated under the Securities Act, the Exchange Act, or any state securities law.

"Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.

"Excluded Registration" means (i) a registration relating to the sale of securities to employees of the Company or a subsidiary of the Company pursuant to a share option, share purchase, or similar plan; (ii) a registration relating to an SEC Rule 145 transaction; (iii) a registration in any form that does not include substantially the same information as would be required to be included in a registration statement covering the sale of the Registrable Securities; or (iv) a registration in which the only Ordinary Shares being registered are Ordinary Shares issuable upon conversion of debt securities that are also being registered.

"Form F-1" means such form under the Securities Act as is in effect on the date hereof or any successor registration form under the Securities Act subsequently adopted by the SEC.

"Form F-3" means such form under the Securities Act as is in effect on the date hereof or any registration form under the Securities Act subsequently adopted by the SEC that permits incorporation of substantial information by reference to other documents filed by the Company with the SEC.

"Holder" means any holder of Registrable Securities.

"Immediate Family Member" means a child, stepchild, grandchild, parent, stepparent, grandparent, spouse, sibling, mother-in-law, father-in-law, son-in-law, daughter-in-law, brother-in-law, or sister-in-law, including, adoptive relationships, of a natural person referred to herein.

"Initiating Holders" means, collectively, Holders who properly initiate a registration request under this Schedule.

"IPO" means the Company's first underwritten public offering of its Ordinary Shares, or the listing for trading of the Company's Ordinary Shares on any recognized stock exchange or regulated market, under the Securities Act or under the equivalent law of another jurisdiction.

"IC" means Israel Corporation Ltd., a company organized under the laws of the State of Israel, and including its successors and permitted assignees.

"Ordinary Shares" means Ordinary Shares of the Company, nominal value NIS 0.03 per share.

"Person" means any individual, corporation, partnership, trust, limited liability company, association or other entity.

"Registrable Securities" means (i) the Ordinary Shares issued to the Original Holders, the Ordinary Shares hereafter acquired by any Original Holder and the Ordinary Shares of any other Person to whom such Ordinary Shares have been transferred in accordance with the terms of this Schedule; excluding in all cases, however, any Ordinary Shares sold by a Person in a transaction in which the applicable rights under this Schedule are not assigned pursuant to Subsection 3.1, and excluding for the purposes of Section 2 any shares for which registration rights have terminated pursuant to Subsection 2.13 of this Schedule; provided that, with respect to any provision of this Schedule that establishes a percentage of Registrable Securities as required to take any action, Registrable Securities shall not include any securities held by the Company or any of its subsidiaries.

"Registrable Securities then outstanding" means the number of shares determined by adding the number of outstanding Ordinary Shares that are Registrable Securities and the number of Ordinary Shares issuable (directly or indirectly) pursuant to any exercisable and/or convertible securities, whether such securities may be exercised or converted immediately or in the future, that are Registrable Securities.

"SEC Rule 144" means Rule 144 promulgated by the SEC under the Securities

"SEC Rule 145" means Rule 145 promulgated by the SEC under the Securities Act.

"Securities Act" means the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.

1

Act.

"Selling Expenses" means all underwriting discounts, selling commissions, and share transfer taxes applicable to the sale of Registrable Securities, and fees and disbursements of counsel for any Holder, except for the fees and disbursements of the Selling Holder Counsel borne and paid by the Company as provided in <u>Subsection 2.7</u>.

- 2. <u>Registration Rights</u>. The Company covenants and agrees as follows:
 - 2.1 <u>IPO and Registration Rights</u>. Subject to determination by the Board of Directors of the Company that a reasonable market capitalisation could be achieved following an IPO, the Company shall use its commercially reasonable efforts to effect an IPO within twenty four (24) months of the date hereof. Following an IPO, the Holders shall have the registration rights set forth below. The rights below are drafted to refer to a registration effected by filing an effective registration statement in compliance with the Securities Act, but are *intended* to apply *mutatis mutandis* in circumstances where an IPO was not effected under the Securities Act but rather under the laws of another jurisdiction as set out in the definition of IPO in Subsection 1.10. In such case, the Company shall be required to take substantially equivalent actions to those described below under the laws of such other jurisdiction.

The Company shall not be liable for any breach of its obligation to use its commercially reasonable efforts pursuant to Clause 2.1, whether in damages or for specific performance, provided such efforts are undertaken in good faith by the Company.

2.2 Demand Registration.

- (a) Form F-1 Demand. If at any time after one hundred eighty (180) calendar days following the effective date of the registration statement for the IPO and until the fifth anniversary thereof, the Company receives a request from Holders of more than twenty percent (20%) of the Registrable Securities then outstanding that the Company file a Form F-1 registration statement with respect to Registrable Securities with a minimum anticipated aggregate offering price, net of Selling Expenses, of Fifteen Million United States Dollars (\$15,000,000), then the Company shall (x) within ten (10) calendar days after the date such request is given, give notice thereof (the "Demand Notice") to all Holders other than the Initiating Holders; and (y) as soon as practicable, and in any event within sixty (60) calendar days after the date such request is given by the Initiating Holders, file a Form F-1 registration statement under the Securities Act covering all Registrable Securities that the Initiating Holders requested to be registered and any additional Registrable Securities requested to be included in such registration by any other Holders, as specified by notice given by each such Holder to the Company within twenty (20) calendar days of the date the Demand Notice is given, and in each case, subject to the limitations of Subsections 2.2(c) and 2.4.
- (b) Form F-3 Demand. As soon as practical after the IPO, but in no event more than eighteen (18) months after the IPO, the Company shall use commercially reasonable efforts in order to qualify for registration on Form F-3 or any comparable or successor form or forms and to maintain such qualification after the Company has qualified for the use of Form F-3. After the Company has qualified for the use of Form F-3 and until five (5) years following an IPO, if the Company receives a request from Holders of at least twenty five percent (25%) of the Registrable Securities then outstanding that the Company file a Form F-3 registration statement with respect to outstanding Registrable Securities of such Holders having an anticipated aggregate offering price, net of Selling Expenses, of at least Five Million United States Dollars (\$5,000,000), then the Company shall (i) within ten (10) calendar days after the date such request is given, give a Demand Notice to all Holders other than the Initiating Holders; and (ii) as soon as

practicable, and in any event within forty-five (45) calendar days after the date such request is given by the Initiating Holders, file a Form F-3 registration statement under the Securities Act covering all Registrable Securities that the Initiating Holders requested to be registered and any additional Registrable Securities requested to be included in such registration by any other Holders, as specified by notice given by each such Holder to the Company within twenty (20) calendar days of the date the Demand Notice is given, and in each case, subject to the availability of a Form F-3 for such offering by the Holders and the limitations of Subsections 2.2(c) and 2.4.

- Notwithstanding the foregoing obligations, if the Company furnishes to Holders requesting a registration pursuant to this Subsection 2.2 a certificate signed by the Company's chief executive officer stating that in the good faith judgment of the Company's Board of Directors it would be materially detrimental to the Company and its shareholders for such registration statement to either become effective or remain effective for as long as such registration statement otherwise would be required to remain effective, because such action would (i) materially interfere with a significant acquisition, corporate reorganization, or other similar transaction involving the Company or (ii) require premature disclosure of material information that the Company has a bona fide business purpose for preserving as confidential; or (iii) render the Company unable to comply with requirements under the Securities Act or Exchange Act, then the Company shall have the right to defer taking action with respect to such filing (and any time periods with respect to filing or effectiveness shall be tolled correspondingly) for a period of not more than forty-five (45) calendar days after the request of the Initiating Holders is given provided that the Company shall not register any securities for its own account or that of any other shareholder during such forty-five (45) calendar day period other than in an Excluded Registration.
- (d) The Company shall not be obligated to effect, or to take any action to effect, any registration pursuant to Subsection 2.2(a) (i) during the period that is sixty (60) calendar days before the Company's good faith estimate of the date of filing of, and ending on a date that is one hundred and eighty (180) calendar days after the effective date of, a Companyinitiated registration of Ordinary Shares, provided that the Company offers to register Registrable Securities in accordance with Subsection 2.3 and the Company is actively employing in good faith commercially reasonable efforts to cause such registration statement to become effective, and the Company shall use its commercially reasonable efforts to cause such registration statement to be declared effective no later than ninety (90) calendar days from the date of any Demand Notice; (ii) after the Company has effected three (3) registrations pursuant to Subsection 2.2(a); or (iii) if the Initiating Holders propose to dispose of Ordinary Shares consisting of Registrable Securities that may be immediately registered on Form F-3 pursuant to a request made pursuant to Subsection 2.2(b). The Company shall not be obligated to effect, or to take any action to effect, any registration pursuant to Subsection 2.2(b) (i) during the period that is thirty (30) calendar days before the Company's good faith estimate of the date of filing of, and ending on a date that is ninety (90) calendar days after the effective date of, a Company-initiated registration of Ordinary Shares, provided that the Company offers to register Registrable Securities in accordance with Subsection 2.3 and the Company is actively employing in good faith commercially reasonable efforts to cause such registration statement to become effective, and the Company shall use its commercially reasonable efforts to cause such registration statement to be declared effective no later than ninety (90) calendar days from the date of any request pursuant to Subsection 2.2(b); or (ii) if the Company has effected one (1) registration pursuant to Subsection 2.2(b) within the twelve (12) month period immediately preceding and the date of such request. A registration shall not be counted as "effected" for purposes of this Subsection 2.2(d) until such time as the applicable registration statement has been declared effective by the SEC, unless the Initiating Holders withdraw their request for such registration,

elect not to pay the registration expenses therefor, and as a result of such non payment forfeit their right to one demand registration statement pursuant to <u>Subsection 2.7</u>, in which case such withdrawn registration statement shall be counted as "effected" for purposes of this Subsection 2.2(d).

2.3 Company Registration. If the Company proposes to register (including, for this purpose, a registration effected by the Company for shareholders other than the Holders) any of its equity securities under the Securities Act in connection with the public offering of such securities solely for cash (other than in an Excluded Registration), the Company shall, at such time, promptly give each Holder notice of such registration. Upon the request of each Holder given within twenty (20) calendar days after such notice is given by the Company, the Company shall, subject to the provisions of Subsection 2.4, cause to be registered all of the Registrable Securities that each such Holder has requested to be included in such registration. The Company shall have the right to terminate or withdraw any registration initiated by it under this Subsection 2.3 before the effective date of such registration, whether or not any Holder has elected to include Registrable Securities in such registration, and any Holder shall have the right to withdraw all or part of its Registrable Securities before the effective date of such registration. The expenses (other than Selling Expenses) of such withdrawn registration shall be borne by the Company in accordance with Subsection 2.7.

2.4 <u>Underwriting Requirements.</u>

- If, pursuant to Subsection 2.2, the Initiating Holders intend to distribute the Registrable Securities covered by their request by means of an underwriting, they shall so advise the Company as a part of their request made pursuant to Subsection 2.2, and the Company shall include such information in the Demand Notice. The underwriter(s) will be selected by the Company and shall be reasonably acceptable to a majority in interest of the Initiating Holders. In such event, the right of any Holder to include such Holder's Registrable Securities in such registration shall be conditioned upon such Holder's participation in such underwriting and the inclusion of such Holder's Registrable Securities in the underwriting to the extent provided herein. All Holders proposing to distribute their securities through such underwriting shall (together with the Company as provided in Subsection 2.5(e)) enter into an underwriting agreement in customary form with the underwriter(s) selected and approved for such underwriting in accordance with Section 2.4. Notwithstanding any other provision of this Subsection 2.4, if the managing underwriter(s) advise(s) the Initiating Holders in writing that marketing factors require a limitation on the number of shares to be underwritten, then the Initiating Holders shall so advise all Holders of Registrable Securities that otherwise would be underwritten pursuant hereto, and the number of Registrable Securities that may be included in the underwriting shall be allocated among such Holders of Registrable Securities, including the Initiating Holders, in proportion (as nearly as practicable) to the number of Registrable Securities owned by each such selling Holder or in such other proportion as shall mutually be agreed to by all such selling Holders; provided, however, that the number of Registrable Securities held by the Holders to be included in such underwriting shall not be reduced unless all securities other than those to be sold by the selling Holders are first entirely excluded from the underwriting. To facilitate the allocation of shares in accordance with the above provisions, the Company or the underwriters may round the number of shares allocated to any Holder to the nearest one hundred (100) shares.
- (b) In connection with any offering involving an underwriting of the Company's share capital pursuant to <u>Subsection 2.3</u>, the Company shall not be required to include any of the Holders' Registrable Securities in such underwriting unless the Holders accept the

terms of the underwriting as reasonably agreed upon between the Company and its underwriters, and then only in such quantity as the underwriters in their sole discretion determine will not jeopardize the success of the offering by the Company. If the total number of securities, including Registrable Securities, requested by shareholders to be included in such offering exceeds the number of securities to be sold (other than by the Company) that the underwriter(s) in their discretion determine is compatible with the success of the offering, then the Company shall be required to include in the offering only that number of such securities, including Registrable Securities, which the underwriters in their sole reasonable discretion determine will not jeopardize the success of the offering. If the underwriter(s) determine that less than all of the Registrable Securities requested to be registered can be included in such offering, then the Registrable Securities that are included in such offering shall be allocated among the selling Holders in proportion (as early as practicable to) the number of Registrable Securities owned by each such selling Holder or in such other proportions as shall mutually be agreed to by all such Notwithstanding the foregoing, in no event shall the number of Registrable Securities included in the offering be reduced unless all securities other than those to be sold by the selling Holders (other than securities to be sold by the Company) are first entirely excluded from the offering. To facilitate the allocation of shares in accordance with the above provisions, the Company or the underwriters may round the number of shares allocated to any Holder to the nearest one hundred (100) shares. For purposes of the provision in this Subsection 2.3(b) concerning apportionment, for any selling Holder that is a partnership, limited liability company, or corporation, the partners, members, retired partners, retired members, shareholders, and Affiliates of such Holder, or the estates and Immediate Family Members of any such partners, retired partners, members, and retired members and any trusts for the benefit of any of the foregoing Persons, shall be deemed to be a single "selling Holder," and any pro rata reduction with respect to such "selling Holder" shall be based upon the aggregate number of Registrable Securities owned by all Persons included in such "selling Holder," as defined in this sentence.

- (c) In the case of an underwritten offering under Subsections 2.2(a) or (b), the price, underwriting discount and other financial terms for the Registrable Securities shall be determined by the Company and in consultation with the Holders of a majority of the Registrable Securities to be sold.
- 2.5 <u>Obligations of the Company.</u> Whenever required under this Section 2 to effect the registration of any Registrable Securities, the Company shall, as expeditiously as reasonably possible:
- (a) prepare and file, if applicable, within the time periods provided by Subsections 2.2(a) and (b), with the SEC a registration statement, including all exhibits and financial statements required under the Securities Act (such registration statement and the prospectus used in connection with such registration statement shall not include the name of any Holder or its ownership interest in the Registrable Securities and/or the Company (as applicable) without the prior written consent of a Holder (unless such Holder's Registrable Securities are included in such registration statement and provided prior notice of the form of disclosure is given to such Holder) (but before filing such registration statement provide the Selling Holder Counsel copies of all documents to be filed and not file such documents to which the Selling Holder Counsel reasonably objects), and use its commercially reasonable efforts to cause such registration statement to become effective no later than ninety (90) calendar days after such registration statement is filed, and keep such registration statement continuously effective for a period of one hundred and twenty (120) calendar days or, if earlier, until the distribution contemplated in the registration statement has been completed; provided, however, that the Company shall not be deemed to have used commercially reasonable efforts to keep such

registration statement continuously effective if the Company voluntarily takes any action or omits to take any action that would result in Holders not being able to offer and sell any Registrable Securities pursuant to such registration statement during the period it is required to be continuously effective unless such action or omission is (i) permitted by the terms of this Schedule, including Section 2.2(c) hereof, or (ii) required by applicable law; and provided further that (i) such one hundred and twenty (120) calendar day period shall be extended for a period of time equal to the period the Holder is not able to sell any securities included in such registration statement as a result of the Company not keeping the registration statement continuously effective, and (ii) in the case of any registration of Registrable Securities on Form F-3 that are intended to be offered on a continuous or delayed basis, subject to compliance with applicable SEC rules, such one hundred and twenty (120) calendar day period shall be extended for up to one hundred twenty (120) calendar days, if necessary, to keep the registration statement effective until all such Registrable Securities are sold;

- (b) prepare and file with the SEC such amendments and supplements to such registration statement, and the prospectus used in connection with such registration statement, as may be necessary to comply with the Securities Act (including using its reasonable commercial efforts to address any comments from the SEC regarding such registration statement) in order to enable the disposition of all securities covered by such registration statement (but before filing such documents provide the Selling Holder Counsel copies of all documents to be filed and not file such documents to which the Selling Holder Counsel reasonably objects);
- (c) furnish to the selling Holders such numbers of copies of the prospectus used in connection with such registration, including any preliminary prospectus, as required by the Securities Act, and such other documents as the Holders may reasonably request in order to facilitate their disposition of their Registrable Securities;
- (d) use its commercially reasonable efforts to register and qualify the securities covered by such registration statement under such other securities or blue-sky laws of such jurisdictions as shall be reasonably requested by the selling Holders; provided that the Company shall not be required to qualify to do business or to file a general consent to service of process in any such states or jurisdictions, unless the Company is already subject to service in such jurisdiction and except as may be required by the Securities Act;
- (e) in the event of any underwritten public offering, enter into customary agreements (including underwriting and indemnification agreements in customary form) and take all such other actions as any selling Holder or the managing underwriter or underwriters, if any, reasonably request in order to expedite or facilitate the registration and disposition of the Registrable Securities;
- (f) use its commercially reasonable efforts to cause all such Registrable Securities covered by such registration statement to be listed on a national securities exchange or trading system and each securities exchange and trading system (if any) on which similar securities issued by the Company are then listed;
- (g) provide a transfer agent and registrar for all Registrable Securities registered pursuant to this Schedule and provide a CUSIP number for all such Registrable Securities, in each case not later than the effective date of such registration;
- (h) promptly make available for inspection by the selling Holders, any managing underwriter(s) participating in any disposition pursuant to such registration statement,

and any attorney or accountant or other agent retained by any such underwriter or selected by the selling Holders, all financial and other records, pertinent corporate documents, and properties of the Company, and cause the Company's officers, directors, employees, and independent accountants to supply all information reasonably requested by any such seller, underwriter, attorney, accountant, or agent, in each case, as necessary or advisable to verify the accuracy of the information in such registration statement and to conduct appropriate due diligence in connection therewith;

- (i) notify each selling Holder, promptly after the Company receives notice thereof, of the time when such registration statement has been declared effective or a supplement to any prospectus forming a part of such registration statement has been filed;
- (j) after such registration statement becomes effective, notify each selling Holder of (i) any request by the SEC that the Company amend or supplement such registration statement or prospectus, (ii) the issuance by the SEC of any stop order suspending the effectiveness of the registration statement or any order by the SEC or any other regulatory authority preventing or suspending the use of any prospectus, (iii) the receipt by the Company of any notification with respect to the suspension of the qualification of the Registrable Securities for offering or sale in any jurisdiction, and (iv) the receipt by the Company of any notification with respect to the initiation or threatening of any proceeding for the suspension of the qualification of the Registrable Securities for offering or sale in any jurisdiction;
- (k) promptly notify the selling Holders when the Company becomes aware of the occurrence of any event as a result of which the applicable registration statement and the prospectus included in such registration statement contains any untrue statement of a material fact or omits to state a material fact necessary to make the statements therein not misleading, or, if for any other reason it shall be necessary during such time period to amend or supplement such registration statement, prospectus or free writing prospectus in order to comply with the Securities Act and, in either case, as promptly as reasonably practicable thereafter, prepare and file with the SEC, and furnish without charge to the selling Holders and the managing underwriter or underwriters, if any, an amendment or supplement to such registration statement or prospectus which shall correct such misstatement or omission or effect such compliance; and
- (l) promptly incorporate in a prospectus supplement or post-effective amendment to the registration statement such information as the selling Holders agree should be included therein relating to the plan of distribution with respect to such Registrable Securities, and make all required filings of such prospectus supplement or post-effective amendment as soon as reasonably practicable after being notified of the matters to be incorporated in such prospectus supplement or post-effective amendment.

In addition, the Company shall ensure that, at all times after any registration statement covering a public offering of securities of the Company under the Securities Act shall have become effective, it shall have in place an insider trading policy that shall provide that the Company's directors may implement a trading program under Rule 10b5-1 of the Exchange Act.

Each Holder agrees that, upon receipt of notice from the Company upon the occurrence of any event of the kind described in section 2.5(j) hereof, such Holder will immediately discontinue disposition of its Registrable Securities under the registration statement until such Holder's receipt of the supplemented prospectus or amended registration statement or until it is advised in writing by the Company that the use of the applicable prospectus may be resumed.

Each Holder covenants and agrees that it will not sell any Registrable Securities under the registration statement until it has received copies of the amendment or supplement to the registration statement or prospectus in accordance with section 2.5(k) hereof.

- 2.6 <u>Furnish Information</u>. It shall be a condition precedent to the obligations of the Company to take any action pursuant to this <u>Section 2</u> with respect to the Registrable Securities of any selling Holder that such Holder shall furnish to the Company such information regarding itself, the Registrable Securities held by it, and the intended method of disposition of such securities as is reasonably required to effect the registration of such Holder's Registrable Securities.
- 2.7 Expenses of Registration. All expenses (other than Selling Expenses) incurred in connection with registrations, filings, or qualifications pursuant to Section 2, including all registration, filing, and qualification fees; printers' and accounting fees; fees and disbursements of counsel for the Company; marketing and road show expenses; and the reasonable fees and disbursements of one counsel for the selling Holders ("Selling Holder Counsel") chosen by Holders of a majority of the Registrable Securities to be sold and consented to by the Company, which consent shall not be unreasonably withheld, shall be borne and paid by the Company; provided, however, that the Company shall not be required to pay for any expenses of any registration proceeding begun pursuant to Subsection 2.2 if the registration request is subsequently withdrawn at the request of the Holders of a majority of the Registrable Securities that were to be registered thereunder (in which case all selling Holders shall bear such expenses pro rata based upon the number of Registrable Securities that were to be included in the withdrawn registration), unless the Holders of a majority of the Registrable Securities agree to forfeit their right to one registration pursuant to Subsections 2.2(a) or 2.2(b), as the case may be then the Holders shall not be required to pay any of such expenses and shall forfeit their right to one registration pursuant to Subsections 2.1(a) or 2.1(b). All Selling Expenses relating to Registrable Securities registered pursuant to this Section 2 shall be borne and paid by the Holders pro rata on the basis of the number of Registrable Securities registered on their behalf.
- 2.8 <u>Delay of Registration</u>. No Holder shall have any right to obtain or seek an injunction restraining or otherwise delaying any registration pursuant to this Schedule as the result of any controversy that might arise with respect to the interpretation or implementation of this Section 2.
- 2.9 <u>Indemnification</u>. If any Registrable Securities are included in a registration statement under this Section 2:
- (a) To the extent permitted by law, the Company will indemnify and hold harmless each selling Holder, and the Affiliates, partners, members, officers, directors, employees, and shareholders of each such Holder; legal counsel and accountants for each such Holder; any underwriter (as defined in the Securities Act) for each such Holder; and each Person, if any, who controls such Holder or underwriter within the meaning of the Securities Act or the Exchange Act, against any Damages, and the Company will pay to each such Holder, underwriter, controlling Person, or other aforementioned Person any legal or other expenses reasonably incurred thereby in connection with investigating or defending any claim or proceeding from which Damages may result, as such expenses are incurred; provided, however, that the indemnity agreement contained in this Subsection 2.8(a) shall not apply to amounts paid in settlement of any such claim or proceeding if such settlement is effected without the consent of the Company, which consent shall not be unreasonably withheld, unless such settlement is entered into without the consent of the Company (i) more than thirty (30) calendar days after

receipt by the Company of a request for reimbursement pursuant to this subsection and (ii) the Company shall not have responded to such request for reimbursement, nor shall the Company be liable for any Damages to the extent that they arise out of or are based upon actions or omissions made in reliance upon and in conformity with written information furnished by or on behalf of any such Holder, underwriter, controlling Person, or other aforementioned Person expressly for use in connection with such registration.

- To the extent permitted by law, each selling Holder, severally and not jointly, will indemnify and hold harmless the Company, and each of its directors, each of its officers who has signed the registration statement, each Person (if any), who controls the Company within the meaning of the Securities Act, legal counsel and accountants for the Company, any underwriter (as defined in the Securities Act), any other Holder selling securities in such registration statement, and any controlling Person of any such underwriter or other Holder, against any Damages, in each case only to the extent that such Damages arise out of or are based upon actions or omissions made in reliance upon and in conformity with written information furnished by or on behalf of such selling Holder expressly for use in connection with such registration; and each such selling Holder will pay to the Company and each other aforementioned Person any legal or other expenses reasonably incurred thereby in connection with investigating or defending any claim or proceeding from which Damages may result, as such expenses are incurred; provided, however, that the indemnity agreement contained in this Subsection 2.8(b) shall not apply to amounts paid in settlement of any such claim or proceeding if ment is effected without the consent of the Holder, which consent shall not be such settle unreasonably withheld, unless such settlement is entered into without the consent of the Holder (i) more than thirty (30) calendar days after receipt by such Holder of a request for reimbursement pursuant to this subsection and (ii) the Holder shall not haveresponded to such request for indemnification; and provided further that in no event shall the aggregate amounts payable by any Holder by way of indemnity or contribution under Subsections 2.8(b) and 2.8(d) exceed the proceeds from the offering received by such Holder (net of any Selling Expenses paid by such Holder), except in the case of fraud or willful misconduct by such Holder.
- Promptly after receipt by an indemnified party under this Subsection 2.9 of notice of the commencement of any action (including any governmental action) for which a party may be entitled to indemnification hereunder, such indemnified party will, if a claim in respect thereof is to be made against any indemnifying party under this Subsection 2.9, give the indemnifying party notice of the commencement thereof. The indemnifying party shall have the right to participate in such action and, to the extent the indemnifying party so desires, participate jointly with any other indemnifying party to which notice has been given, and to control the defense thereof with counsel mutually satisfactory to the indemnifying and indemnified parties; provided, however, that an indemnified party (together with all other indemnified parties that may be represented without conflict by one counsel) shall have the right to retain one separate counsel and one local counsel, with the fees and expenses to be paid by the indemnifying party, if representation of such indemnified party by the counsel retained by the indemnifying party would be inappropriate due to actual or potential differing interests between such indemnified party and any other party represented by such counsel in such action. The failure to give notice to the indemnifying party within a reasonable time of the commencement of any such action shall not relieve such indemnifying party of any liability to the indemnified party under this Subsection 2.9, except to the extent that such failure materially prejudices the indemnifying party's ability to defend such action (through the forfeiture of substantive rights and defenses). The failure to give notice to the indemnifying party will not relieve it of any liability that it may have to any indemnified party otherwise than under this <u>Subsection 2.9</u>.

- To provide for just and equitable contribution to joint liability under the Securities Act in any case in which either: (i) any party otherwise entitled to indemnification hereunder makes a claim for indemnification pursuant to this Subsection 2.9 but it is judicially determined (by the entry of a final judgment or decree by a court of competent jurisdiction and the expiration of time to appeal or the denial of the last right of appeal) that such indemnification may not be enforced in such case, notwithstanding the fact that this Subsection 2.9 provides for indemnification in such case, or (ii) contribution under the Securities Act may be required on the part of any party hereto for which indemnification is provided under this Subsection 2.9, then, and in each such case, such parties will contribute to the aggregate losses, claims, damages, liabilities, or expenses to which they may be subject (after contribution from others) in such proportion as is appropriate to reflect the relative fault of each of the indemnifying party and the indemnified party in connection with the statements, omissions, or other actions that resulted in such loss, claim, damage, liability, or expense, as well as to reflect any other relevant equitable considerations. The relative fault of the indemnifying party and of the indemnified party shall be determined by reference to, among other things, whether the untrue or allegedly untrue statement of a material fact, or the omission or alleged omission of a material fact, relates to information supplied by the indemnifying party or by the indemnified party and the parties' relative intent, knowledge, access to information, and opportunity to correct or prevent such statement or omission; provided, however, that, in any such case (x) no Holder will be required to contribute any amount in excess of the public offering price of all such Registrable Securities offered and sold by such Holder pursuant to such registration statement, and (y) no Person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) will be entitled to contribution from any Person who was not guilty of such fraudulent misrepresentation; and provided further that in no event shall a Holder's liability pursuant to this Subsection 2.8(d), when combined with the amounts paid or payable by such Holder pursuant to Subsection 2.8(b), exceed the proceeds from the offering received by such Holder (net of any Selling Expenses paid by such Holder), except in the case of willful misconduct or fraud by such Holder.
- (e) Notwithstanding the foregoing, to the extent that the provisions on indemnification and contribution contained in the underwriting agreement entered into in connection with the underwritten public offering are in conflict with the foregoing provisions, the provisions in the underwriting agreement shall control.
- (f) Unless otherwise superseded by an underwriting agreement entered into in connection with the underwritten public offering, the obligations of the Company and Holders under this <u>Subsection 2.9</u> shall survive the completion of any offering of Registrable Securities in a registration under this <u>Section 2</u>, and otherwise shall survive the termination of the provisions in this Schedule.
- 2.10 <u>Reports Under Exchange Act</u>. With a view to making available to the Holders the benefits of SEC Rule 144 and any other rule or regulation of the SEC that may at any time permit a Holder to sell securities of the Company to the public without registration or pursuant to a registration on Form F-3, the Company shall:
- (a) make and keep available adequate current public information, as those terms are understood and defined in SEC Rule 144, at all times after the effective date of the registration statement filed by the Company for the IPO;
- (b) use commercially reasonable efforts to file with the SEC in a timely manner all reports and other documents required of the Company under the Securities Act and the

Exchange Act (at any time after the Company has become subject to such reporting requirements); and

- (c) furnish to any Holder, so long as the Holder owns any Registrable Securities, forthwith upon request (i) to the extent accurate, a written statement by the Company that it has complied with the reporting requirements of SEC Rule 144 (at any time after ninety (90) calendar days after the effective date of the registration statement filed by the Company for the IPO), the Securities Act, and the Exchange Act (at any time after the Company has become subject to such reporting requirements), or that it qualifies as a registrant whose securities may be resold pursuant to Form F-3 (at any time after the Company so qualifies); (ii) a copy of the most recent annual or quarterly report of the Company and such other reports and documents so filed by the Company; and (iii) such other information as may be reasonably requested in availing any Holder of any rule or regulation of the SEC that permits the selling of any such securities without registration (at any time after the Company has become subject to the reporting requirements under the Exchange Act) or pursuant to Form F-3 (at any time after the Company so qualifies to use such form).
- 2.11 <u>Limitations on Subsequent Registration Rights</u>. From and after the Restructuring Effective Time, the Company shall not, without the prior written consent of the Holders of a majority of the Registrable Securities then outstanding, enter into any agreement with any holder or prospective holder of any securities of the Company that would provide to such holder the right to include securities in any registration on terms other than either a pro rata basis with respect to the Registrable Securities or on a subordinated basis after all Holders have had the opportunity to include in the registration and offering all shares of Registrable Securities that they wish to so include.
- "Market Stand-off" Agreement. In the event of a sale by the Company 2.12 of the Company's equity securities in an underwritten offering, each Holder hereby agrees, if requested in writing by the managing underwriter in such underwritten offering, that it will not, without the prior written consent of such managing underwriter, during the period commencing on the date of the final prospectus relating to the registration by the Company of Ordinary Shares or any other equity securities under the Securities Act on a registration statement on Form F-1 or Form F-3, and ending on the date specified by the Company and the managing underwriter (such period not to exceed ninety (90) calendar days, or such other period as may be requested by the Company or an underwriter to accommodate regulatory restrictions) (or one hundred eighty (180) calendar days in the case of the IPO, or such other period as may be requested by the Company or an underwriter to accommodate regulatory restrictions (), (i) lend; offer; pledge; sell; contract to sell; sell any option or contract to purchase; purchase any option or contract to sell; grant any option, right, or warrant to purchase; or otherwise transfer or dispose of, directly or indirectly, any Ordinary Shares or any securities convertible into or exercisable or exchangeable (directly or indirectly) for Ordinary Shares held immediately before the effective date of the registration statement for such offering or (ii) enter into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of such securities, whether any such transaction described in clause (i) or (ii) above is to be settled by delivery of Ordinary Shares or other securities, in cash, or otherwise. The foregoing provisions of this Subsection 2.12 shall not apply to the sale of any shares to an underwriter pursuant to an underwriting agreement, or the transfer of any shares to any Affiliate or any trust for the direct or indirect benefit of the Holder or an Immediate Family Member of the Holder, provided that the Affiliate or trustee of the trust agrees to be bound in writing by the restrictions set forth herein, and provided further that any such transfer to a trust shall not involve a disposition for value, and provided further that any such limitation shall be applicable to the Holders only if the Company's

Affiliates and all of their officers and directors are subject to the same restrictions (and if the any such parties are released by the managing underwriters the Holders are similarly released) and the Company uses commercially reasonable efforts to obtain a similar agreement from all shareholders individually owning at least one percent (1%) of the Company's outstanding Ordinary Shares (after giving effect to conversion into Ordinary Shares of all outstanding preferred shares or other convertible securities). The underwriters in connection with such registration are intended third-party beneficiaries of this Subsection 2.12 and shall have the right, power and authority to enforce the provisions hereof as though they were a party hereto. Each Holder further agrees to execute such agreements as may be reasonably requested by the underwriters in connection with such registration that are consistent with this Subsection 2.12 or that are necessary to give further effect thereto. Any discretionary waiver or termination of the restrictions of any or all of such agreements by the Company or the underwriters shall apply pro rata to all Holders subject to such agreements, based on the number of shares subject to such agreements.

- 2.13 <u>Termination of Registration Rights</u>. The right of any Holder to request registration or inclusion of Registrable Securities in any registration pursuant to <u>Subsections 2.2</u> or <u>2.3</u> shall terminate upon the first to occur of:
- (a) the Company completing its IPO and becoming subject to the provisions of the Exchange Act whereby SEC Rule 144 or another similar exemption under the Securities Act is available for the sale of all of such Holder's shares; and
 - (b) the fifth anniversary of the IPO.

3. Miscellaneous.

- 3.1 Successors and Assigns. The rights set out in this Schedule may be assigned (but only with all related obligations) by a Holder to a transferee of Registrable Securities; provided, however, that (x) the Company is, within a reasonable time after such transfer, furnished with written notice of the name and address of such transferee and the Registrable Securities with respect to which such rights are being transferred; and (y) such transferee agrees in a written instrument delivered to the Company to be bound by and subject to the terms and conditions of this Schedule, including the provisions of Subsection 2.12. For the purposes of determining the number of shares of Registrable Securities held by a transferee, the holdings of a transferee (1) that is an Affiliate or stockholder of a Holder; (2) who is a Holder's Immediate Family Member; or (3) that is a trust for the benefit of an individual Holder or such Holder's Immediate Family Member shall be aggregated together and with those of the transferring Holder; provided further that all transferees who would not qualify individually for assignment of rights shall have a single attorney-in-fact for the purpose of exercising any rights, receiving notices, or taking any action under this Schedule. The terms and conditions of this Schedule inure to the benefit of and are binding upon the respective successors and permitted assignees of the parties. Nothing in this Schedule, express or implied, is intended to confer upon any party other than the parties hereto or their respective successors and permitted assignees any rights, remedies, obligations or liabilities under or by reason of this Schedule, except as expressly provided herein.
- 3.2 <u>Governing Law. Jurisdiction.</u> Notwithstanding clause 22 of the Global Restructuring Deed and except insofar as it relates to provisions of US securities laws, federal or state, which shall be determined by the applicable US law this Schedule shall be governed by, and construed in accordance with, the laws of the State of Israel without regard to the choice of law

principles thereof. Each of the parties hereto irrevocably submits to the exclusive jurisdiction of the courts of Tel Aviv for the purpose of any suit, action, proceeding or judgment relating to or arising out of this Schedule and the transactions contemplated hereby.

- 3.3 <u>Titles and Subtitles</u>. The titles and subtitles used in this Schedule are for convenience only and are not to be considered in construing or interpreting this Schedule.
- Notices. All notices and other communications given or made pursuant to this Schedule shall be in writing and shall be deemed effectively given upon the earlier of actual receipt or (i) personal delivery to the party to be notified; (ii) when sent, if sent by electronic mail or facsimile during the recipient's normal business hours, and if not sent during normal business hours, then on the recipient's next business day; (iii) five (5) calendar days after having been sent by registered or certified mail, return receipt requested, postage prepaid; or (iv) one (1) business day after the business day of deposit with a nationally recognized overnight courier, freight prepaid, specifying next-day delivery, with written verification of receipt. All communications shall be sent to the Holders to their respective addresses as identified in the signature pages to the Global Restructuring Deed (of which this Schedule forms a part), or to the principal office of the Company and to the attention of the Chief Executive Officer, in the case of the Company, or to such email address, facsimile number, or address as subsequently modified by written notice given in accordance with this <u>Subsection 3.4</u>.
- Amendments and Waivers. Any term of this Schedule may be amended 3.5 or restated and the observance of any term of this Schedule may be waived (either generally or in a particular instance, and either retroactively or prospectively) only with the written consent of the Company and Holders of a 75 per cent. majority of the Registrable Securities then outstanding, provided that the amendment, restatement or waiver does not unfairly discriminate against the non-consenting Holders; provided further that any right granted hereunder may be waived by the party holding such right, without the consent of any other party. Notwithstanding the foregoing, this Schedule may not be amended, restated or terminated and the observance of any term hereof may not be waived with respect to any Holder without the written consent of such Holder, unless such amendment, termination, or waiver applies to all Holders in the same fashion. The Company shall give prompt notice of any amendment or termination hereof or waiver hereunder to any party hereto that did not consent in writing to such amendment, termination, or waiver. Any amendment, termination, or waiver effected in accordance with this Subsection 3.5 shall be binding on all parties hereto, regardless of whether any such party has consented thereto. No waivers of or exceptions to any term, condition, or provision of this Schedule, in any one or more instances, shall be deemed to be or construed as a further or continuing waiver of any such term, condition, or provision.
- 3.6 <u>Severability</u>. In case any one or more of the provisions contained in this Schedule is for any reason held to be invalid, illegal or unenforceable in any respect, such invalidity, illegality, or unenforceability shall not affect any other provision of this Schedule, and such invalid, illegal, or unenforceable provision shall be reformed and construed so that it will be valid, legal, and enforceable to the maximum extent permitted by law.
- 3.7 <u>Aggregation of Shares</u>. All shares of Registrable Securities held or acquired by Affiliates shall be aggregated together for the purpose of determining the availability of any rights under this Schedule and such Affiliated persons may apportion such rights as among themselves in any manner they deem appropriate.

- 3.8 <u>Entire Agreement</u>. This Schedule constitutes the full and entire understanding and agreement among the parties with respect to the subject matter hereof, and any other written or oral agreement relating to the subject matter hereof existing between the parties is expressly canceled.
- 3.9 <u>Delays or Omissions</u>. No delay or omission to exercise any right, power, or remedy accruing to any party under this Schedule, upon any breach or default of any other party under this Schedule, shall impair any such right, power, or remedy of such nonbreaching or nondefaulting party, nor shall it be construed to be a waiver of or acquiescence to any such breach or default, or to any similar breach or default thereafter occurring, nor shall any waiver of any single breach or default be deemed a waiver of any other breach or default theretofore or thereafter occurring. All remedies, whether under this Schedule or by law or otherwise afforded to any party, shall be cumulative and not alternative.
- 3.10 <u>Injunctive Relief.</u> Save in respect of the Company's obligation to use commercially reasonable efforts in Clause 2.1, it is hereby agreed and acknowledged that it will be impossible to measure in money the damage that would be suffered if the parties fail to comply with any of the obligations herein imposed on them and that in the event of any such failure, an aggrieved Person will be irreparably damaged and will not have an adequate remedy at law. Any one or more Holders of at least 10 percent. of the Registrable Securities then outstanding shall, therefore, be entitled (in addition to any other remedy to which it may be entitled in law or in equity) to injunctive relief, including specific performance, to enforce such obligations, and if any action should be brought in equity to enforce any of the provisions of this Schedule, none of the parties hereto shall raise the defense that there is an adequate remedy at law.

[Remainder of Page Intentionally Left Blank]

SCHEDULE 16

INITIAL DIRECTORS AND CHAIRMAN⁴⁵

1. Initial Directors as at the Restructuring Effective Time

Name of Director		Passport Number	Term of office of Directorship
1.			3 years
<u>12.</u>	Shlomo Sharon		2 years
<u>2</u> 3.	Regina Ungar		3 years
4.			2 years
<u>3</u> <u>5.</u>	Fleming Jacobs		3 years
	(An Interim Appointee, in	<u>f appointed)</u>	(If Interim Appointee, until the appointment of the Director/Chairman in accordance with the articles of association of the Company)
<u>4</u> 6.	Dr Karsten Liebing		2 years
	(An Interim Appointee, in	f appointed)	(If Interim Appointee, until the appointment of the Director/Chairman in accordance with the articles of association of the Company)

⁴⁵ Note: The appointment of the Initial Directors and the Additional Initial Director—is subject to finalising the terms and conditions of their employment.

I

Name of Director		Passport Number	Term of office of Directorship
5	Rafi Danieli		Until terminated in accordance with the articles of association of the Company applicable to IC Directors in effect following the Restructuring Effective Time and as subsequently amended from time to time
6	Yoav Zeva		Until terminated in accordance with the articles of association of the Company applicable to IC Directors in effect following the Restructuring Effective Time and as subsequently amended from time to time
<u>77.</u>	Dimitris Chatzis		3 years
<u>8 8.</u>	Peter Birger Meyer		2 years
9.			3 years
9 [], Chairman		of the Board to be appointed articles of association of the	3 years (as Director)

2.Nomination of Additional Initial Director after the Restructuring Effective Time

The Shipowners listed at paragraphs 1 to 19 inclusive (but excluding paragraph 17) of Part IV (Shipowners) of Schedule 1 (Obliging and Participating Stakeholders) to this Deed (the

"Nominating Entities") will together have a right to nominate one Director to the Board of Directors of the Company within the 6 weeks period after the Restructuring Effective Time, by providing to the Company a written notice which will be delivered by an authorised signatory for and on behalf of Conti Group as an authorised representative of the Nominating Entities and shall constitute a binding nomination notice on behalf of the Nominating Entities.

I